Contents lists available at ScienceDirect



Renewable and Sustainable Energy Reviews

journal homepage: www.elsevier.com/locate/rser



# Do concerns about wind farms blow over with time? Residents' acceptance over phases of project development and proximity



Julia le Maitre<sup>a,b,\*</sup>, Geraldine Ryan<sup>a,b</sup>, Bernadette Power<sup>a,b</sup>

<sup>a</sup> Cork University Business School, University College Cork, Cork, Ireland

<sup>b</sup> Environmental Research Institute, University College Cork, Cork, Ireland

#### ARTICLE INFO

#### JEL classification: D62 D80 D90 Q20 Q40 Q50 Keywords: Wind energy Willingness to accept Experience Project development Public engagement Proximity Latent class Community acceptance

# ABSTRACT

Social acceptance is a key issue for the continued expansion of onshore wind energy. Wind energy development targets increasingly rely on the assumption that residents' concerns related to new wind farms dissipate over time. The persistence of resistance to new wind farms has motivated efforts to investigate this effect. The 'Ucurve' hypothesis proposes that acceptance is likely to decrease when residents are confronted with the planning of a wind farm in their neighbourhood, but that acceptance may later recover during construction and operation. In this study, relevant research is reviewed, discussed, and applied using a largescale experimental survey focused on residents living within 10 km of an existing wind farm in Ireland (n = 1109). It uses two indicators of how people experience wind farms to investigate willingness to accept further developments. The indicators include the proximity of existing wind farms and their development phase (i.e., planning, construction or operation). The findings show that experience is an important determinant of acceptance, as are an awareness of low-carbon energy initiatives and sense of community spirit. The study examines residents' expectations for participatory fairness and local benefits. Expected adverse impacts on local tourism or potential for discord within the community influence the acceptance for further development. Acceptance is also determined by trust in sources of information, including a designated community liaison officer. The concerns of residents living within the nearest 2 km radius of a wind farm and at the earliest and most uncertain phases of project planning can be crucial issues for acceptance.

# 1. Introduction

Living close to a wind farm is expected to become an increasingly common experience for non-urban communities in regions with high wind potential [1,2]. Wind energy has established itself as a pre-eminent technology in the transition to renewable energy sources and its accelerated development is anticipated to add 570 GW worldwide by 2027 [3,4]. Therefore, a critical assumption for expanding onshore wind capacity is that local residents will become accustomed to wind farm developments as they continue to be developed near residential areas [2, 5]. However, evidence of whether and how acceptance of wind farms progresses during implementation has been limited to date [5]. The operational lifetime of a wind farm provides evident national benefits in the form of affordable renewable electricity without producing greenhouse gas emissions [6]. Nonetheless, the social impacts of wind farms disproportionately affect the immediate surrounds of the project. These impacts persist until project decommissioning or, increasingly, repowering [7,8]. Acceptance for local energy transitions is therefore shaped over the lifetime of renewable energy infrastructure. This relationship can be described as a U-shaped curve, which hypothesises that residents' attitudes towards wind farms are likely to be less favourable during the planning phase of development but may gradually become more favourable once a wind farm enters into operation [9]. Understanding this process calls for deeper investigation as to how the process of acceptance is affected by prior experience [10]. The dynamics of social acceptance of renewable energy has become a high-priority research focus [11,12] and the topic of a recent special issue [13]. To contribute towards closing this gap, the paper investigates how willingness to accept further wind farm development progresses over different phases of project development for residents living within 2 km, 2-5 km or 5-10 km of a wind farm. It investigates how this process is influenced by the individual-specific trade-offs associated with wind farm proposals. It also profiles respondents whose acceptance varies with prior involvement with renewables, trust in information about a

\* Corresponding author. Cork University Business School, University College Cork, Cork, Ireland. *E-mail address*: jlemaitre@ucc.ie (J. le Maitre).

https://doi.org/10.1016/j.rser.2023.113839

Received 11 March 2023; Received in revised form 15 September 2023; Accepted 3 October 2023 Available online 21 October 2023 1364-0321/© 2023 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/). new wind farm proposal, and attitudes towards onshore wind-generated electricity.

It has been suggested that residents grow accustomed to wind farms during the development process and that this contributes to acceptance, whether due to the realisation of local benefits of wind farms [14] or with first-hand experience (rather than the expectation) of wind farm impacts [15]. Various studies have highlighted prior experience of renewable energy as an important consideration for the local acceptance of new developments [1,2,5,8,14,15]. Survey methods are helpful to investigate the relationship between wind energy development and local attitudes relevant to planning for wind farm implementation which considers local concerns [2]. An important inclusion in such methods is to consider variables for residents' "experience" of wind farm developments within the local setting using a Euclidian proximity measure, such as the distance and/or view of the turbines from their residence ([2] p.3). However, the relationship between prior experience of wind farm developments and local attitudes has an important temporal dimension [5,8,12].

In the short term, the tendency to preserve the status quo when confronted with an uncertain imagined future for an environment can heighten initial resistance to a wind farm development [16,17] and prior attitudes might never return to the same level [7,18]. It is essential that these possible outcomes be better understood and incorporated into planning and policy. However, research into how social responses towards renewable energy change over time is scarce, owing to conceptual challenges [19] as well as imperfect empirical assessment methodologies [20]. There has been a significant interdisciplinary effort to understand how familiarity, knowledge and experiences of nearby wind farms affect the acceptance of wind energy [2,15,21,22,23,24]. This paper makes headway towards this objective and uses two indicators of residents' experiences of wind energy to investigate whether and how acceptance develops with greater/longer-term exposure to wind farm developments. It considers residents' experiences of wind farms owing to exposure to existing projects over (i) the temporal phases of project development (planning, construction and operation); and (ii) by residents' proximity to the wind farm (within 2 km, 2-5 km or 5-10 km). These effects on local acceptance are investigated under the research question, "Do residents who live nearer to wind farms, or with a longer exposure to wind farms, have stronger acceptance for further development in their community?" Addressing this question contributes to a fuller understanding of residents' responses towards renewable energy development in the community. Across wind-rich areas, it is becoming more urgent to understand stakeholders' prevailing concerns if the energy transition is to go forward equitably and at the necessary pace [25,26].

Renewable energy developments affect diverse interests [27]. Acceptance is moderated by experience and knowledge of renewable energy innovation which has an ostensibly different influence on stakeholders' perception of risks, costs, benefits and impression of fairness during the phases of the development process [21]. Long-term attitudes can be shaped by residents' perception of the distributive and procedural justice of wind farm developments throughout planning, construction and operation [5,25,21]. For example, residents may place importance on the socioeconomic or environmental outcomes for communities neighbouring a wind farm [7]. Their expectations for fairness can be affected by their underlying trust in information sources about the development and the actors involved in its implementation [28,29]. Therefore, information, involvement, and stakeholders' desired outcomes have been identified as key concerns for the local acceptance of renewable energy technologies [11,30]. This study aims to identify segments of respondents who might benefit from common interventions during the development process based on individual-specific concerns,

priorities and attitudes towards wind development. A latent class approach is used to examine the factors affecting wind farm neighbours' willingness to accept further wind development in the community. Therefore, the paper also addresses the following question: "To what extent do community/individual circumstances, attitudes, and awareness of renewable energy influence wind farm neighbours' willingness to accept further development in the area?" Characterising how experiences of renewable energy, perceived costs/benefits and personal norms determine willingness to accept accelerated onshore wind is essential for the shift towards sustainable electricity use. Policy is increasingly called upon to recognise that stakeholder confidence, awareness and involvement can act as "social catalysts" or "soft barriers" for/against the diffusion of renewables [31,32].

To our knowledge, this paper is the first largescale choice experiment examining the relationship between residents' acceptance for further wind development and their experience of wind energy implementation temporally, across each phase of wind farm development, with sampling based on geographic proximity to the wind farm. The paper investigates the acceptance of Irish citizens living within 10 km of a wind farm and examines their preferences for further wind energy developments located in the community. Ireland is an illustrative case/site to study the experience of communities near to wind farms on account of its vast onshore wind potential [33], small and rural population, and the increasing pressure on its onshore wind farm planning system [34]. This adds to comparable studies which have been conducted at a national scale (e.g., Refs. [35,36-39]), but which do not study respondents' experience of wind farms within their area of residence. It also identifies and characterises unobserved latent factors that shape the conditional acceptance of onshore wind farm development at the individual level [40]. This research contributes towards Sustainable Development Goals 7, 11 and 13 relating to fair infrastructure development processes in renewable energy for climate action. Section 2 provides an overview of the theory concerning wind farm neighbours' experiences and attitudes towards wind farm developments. Section 3 details the methodology. Section 4 contains the results and discussion. Section 5 presents implications for policy and future research.

# 2. Theory

Society's increasing reliance on wind electricity will continue to require the deployment of numerous wind farms across predominantly rural areas [41–43][59]. This implies that residents in these areas are more likely to encounter and become accustomed to onshore wind farms through their daily activities in a way not seen for dispatchable energy sources which supply electricity to the centralised system from largescale power plants. However, research to investigate whether prior experience of wind farms leads to greater tolerance or support for further development has been limited [5]. There is a growing body of evidence which highlights the planning phase as a crucial point for social acceptance [5,7,29,9,44,45]. During planning, hesitancy to accept uncertain changes from the status quo [5,46] can manifest as resistance towards wind energy developments or fear about the impacts of a proposed project [1,5,11,9]. However, equally important for the energy transition is the nature of acceptance beyond this point, and the ways in which residents continue to interact with, and respond to, onshore wind energy developments [5,23,47]. Acceptance is characterised by dynamic interactions between society and technology which ultimately shape the perception of the justice of new developments [9,48,49]. However, residents may feel alienated by shortcomings in the distribution of benefits between developers from outside the community and local inhabitants [5,7],42,50 or if the participation and representation of ω

Empirical research into the effects of experience on the local acceptance of onshore wind farms.

| Study                              | Measure of experience   | Interval   | Location (Commercial operation date)   | Project scale (Total capacity)   | Sample  | Explanatory factors   | Effect on<br>local<br>acceptance |
|------------------------------------|---|--|--|--|---|---|----------------------------------|
| Warren <i>et al.</i><br>(2005)     | Proximity<br>Pre-/post-<br>construction                               | 0–5 km, 5–10 km,<br>10–20 km   | Scottish Borders:<br>Dun Law, Scotland (2000)<br>Black Hill, Scotland (1999)<br>South-West Ireland:<br>Currabwee, (2000)<br>Milane Hill (2000)<br>Beenageeha (2000)<br>Tuarsillagh (2000)  | 26 turbines, 42 m (17.2<br>MW)<br>22 turbines, 50 m (22<br>MW)<br>8 turbines, 60 m (4.8<br>MW)<br>9 turbines, 60 m (4.8<br>MW)<br>6 turbines, 47 m (5.94<br>MW)<br>6 turbines, 47 m (3.96<br>MW)<br>23 turbines, 50 m (15.8<br>MW) | Scottish Borders (115)<br>South-West Ireland (240)  | Visual factors, local benefits, tourism, noise<br>annoyance, intrinsic value, siting preferences,<br>attitude towards first/second wind farm proposal   | +                                |
| Wolsink<br>(2007)                  | Pre-planning,<br>concrete planning<br>stage and post-<br>construction | 16 years (16 public<br>survey occasions)                             | Multiple locations   | Multiple projects and locations  | 1733 Wadden Vereniging<br>environmental protection survey<br>respondents (pre-planning: 499;<br>concrete planning stage: 554; post-<br>construction: 680) | Perceptions of visual changes to the landscape,<br>processes for citizen involvement during the<br>planning/siting negotiation and development<br>process   | +                                |
| Eltham <i>et al.</i><br>(2008)     | Pre-/post-<br>construction  | 14 years (recalled opinion)  | St Newland East village<br>within 2.25 km of Carland<br>Cross, Cornwall, UK (1992)   | 15 turbines, 30 m (6 MW)   | 100 (semi-structured interviews)  | Visual factors, noise annoyance, technical or<br>contextual disruption, tourism impacts,<br>environmental benefits, financial benefits  | +                                |
| Swofford and<br>Slattery<br>(2010) | Proximity   | 0–5 km, 5–10 km,<br>10–20 km   | Wolf Ridge Wind Farm,<br>Cooke County, Texas, US<br>(2008)   | 75 turbines, 80 m (112.5<br>MW)  | 200 surveys (8 <5 km; 106 5–10<br>km; 86 10–20 km)  | Visual impact, noise annoyance, property value-<br>loss, reliability, necessity/cost of reducing fossil<br>fuel use, global environmental benefits, areas<br>where wind farms are visible   | _                                |
| Baxter et al.<br>(2013)            | Proximity   | Comparative case-<br>control survey within<br>15 km                  | Cases: Melancthon I (2006)<br>and II (2008), Ontario,<br>Canada.<br>Control: West Perth,<br>Ontario, Canada (no<br>turbines)   | Melancthon I: 45 turbines<br>(67.5 MW)<br>Malancthon II: 88<br>turbines (132 MW)   | 300 surveys within 15 km  | Self-assessed knowledge and preferences for<br>renewable energy, visual factors, health impacts,<br>biodiversity impacts, economic impacts, siting<br>process, community conflict, opposition to existing<br>infrastructure   | +                                |
| Meyerhoff<br>(2013)                | Proximity   | Local/global<br>autocorrelation within<br>a distance band of 6<br>km | Westsachsen, Saxony,<br>Germany  | Multiple projects  | 353 telephonic interviews   | Local/global autocorrelation within a distance<br>band of 6 km, Size of wind farms, maximum height<br>of turbines, minimum distance to residential areas,<br>monthly surcharge to electricity bill, effect on red<br>kite bird population   | +                                |
| Rijnsoever<br>et al. (2015)        | Pre-/post-<br>Fukushima nuclear<br>accident                           | 2 years (unpaired sample)  | Households across the<br>Utrecht province,<br>Netherlands  | Multiple survey locations  | 916 (2010) and 1448 (2012) hand-<br>delivered surveys   | Un/labelled energy source, security of supply,<br>spatial impact, amount of effort, price per kilowatt<br>hour, long-term problems  | -                                |
| Motosu and<br>Maruyama<br>(2016)   | Proximity   | 0–1 km/1–2 km/2–3<br>km  | Within 3 km of X Wind Farm,<br>Japan   | Confidential project   | 298 paper surveys   | Global environmental benefits, energy security,<br>local benefits, creation of a local image, creation of<br>a new landscape, tourism, noise annoyance,<br>shadow flicker, light obstruction, sources of<br>information about the wind farm, trust in the<br>developer. developer responses to concerns | -                                |
| Petrova et al.<br>(2016)           | Proximity   | 0.25–0.5 mi, 0.5–0.75<br>mi, 0.75–1 km                               | Hull Wind I (2001) and Wind<br>II (2006), Plymouth County,<br>Massachusetts, USA<br>Kingston wind turbines<br>(2011–2012), Plymouth<br>County, Massachusetts, USA<br>Falmouth Wind I and II<br>(2010), Barnstable County,<br>USA | Hull Wind I: 1 turbine<br>(660 kW), Hull Wind II: 1<br>turbine (1.8 MW)<br>Kingston: 3 private<br>turbines (6 MW),<br>transport authority (100<br>kW), 1 municipal turbine<br>(2 MW)<br>Falmouth Wind I: 1                         | 345, 350 and 356 surveys collected<br>within each town respectively   | Visual/landscape factors, noise annoyance,<br>environmental concern, socioeconomic benefits,<br>procedural aspects (VESPA)  | +/-                              |

| מחזב ז (רחווווות              | ( nc                       |  |  |   |  |  |                                  |
|-------------------------------|----------------------------|--|--|---|--|--|----------------------------------|
| Study                         | Measure of<br>experience   | Interval                                     | Location (Commercial<br>operation date)  | Project scale (Total<br>capacity)   | Sample   | Explanatory factors  | Effect on<br>local<br>acceptance |
|                               |                            |  |  | turbine (1.25 MW),<br>Falmouth Wind II: 1<br>turbine (1.25 MW)  |  |  |                                  |
| Wilson and<br>Dyke (2016)     | Pre-/post-<br>construction | 5 years (recalled<br>opinion)                | Within 3 km radius of<br>Roskrow Barton, Cornwall,<br>UK (2008)  | 2 turbines, 70 m (1.7<br>MW)  | 58 key stakeholders (semi-<br>structured interviews) | Place attachment, visual impact, noise annoyance,<br>environmental impacts, changes to local area,<br>trafific congestion during construction, property<br>value-loss      | -/+                              |
| Hoen <i>et al.</i><br>(2019)  | Proximity                  | <0.8 km, 0.8–1.6 km,<br>1.6–4.8 km, 4.8–8 km | Within 8 km of a wind<br>turbine, US   | Multiple projects   | 1705 (online survey)                                 | Auditory and visual sensory perception, perceived<br>planning process fairness, demographic<br>characteristics, environmental concern                                      | +                                |
| Mills <i>et al.</i><br>(2019) | Post-construction          | 2 years (individually<br>repeated survey)    | Michigan 1, Michigan<br>(2008)<br>Harvest, Michigan (2008/<br>2012)<br>Sigel, Michigan (2012)<br>Stoney Corners, Michigan<br>(2008/2010) | 46 turbines, 119 m<br>65 turbines, 121/145 m<br>40 turbines, 150 m<br>29 turbines, 126/146 m  | 520 landowners (online surveys)<br>149/172/126/73    | Visual impact, noise annoyance, local benefits,<br>biodiversity impacts, procedural justice, health<br>impacts, perceived changes in local weather,<br>property value-loss | -×+                              |
| Windemer,<br>(2023)           | End-of-life                | 25 years (recalled opinion)                  | Within 3.5 km of St Breock,<br>Cornwall (1994) and Kirkby<br>Moor, Lake District National<br>Park (1993) UK                              | 12 turbines, 42.4 m;<br>repowering application<br>refused<br>11 turbines, 53.5 m,<br>repowering in 2015<br>received wide support,<br>647.5 turbines 100 m | Kirkby Moor: 128 (12.5 MW)<br>St Breock: 74 (18 MW)  | Awareness of 25-year planning consent,<br>community involvement in the planning process,<br>local benefits, place attachments, environmental<br>concern                    | -/+                              |

residents in decisions affecting the community are perceived to be undemocratic [5,7,25,29,51]. This section provides an overview of relevant empirical research investigating the ways in which residents "get used to" wind farms [5].

Table 1 provides an overview of thirteen studies which have considered how stakeholders' level of experience of onshore wind farm development affects local acceptance. One strand of research gives greater emphasis to stakeholders' temporal exposure to wind farms as a measure and aims to understand whether there are activities at specific phases of the wind farm development process which are most salient for local support. Another strand focuses on proximity to wind farms and aims to explore which conditions of new developments are most conducive to the acceptance of wind farms sited within the community. The table's chronological ordering reflects the evolution of research into the diffusion of wind energy technology over the past two decades. This research progression has been reviewed in detail by Batel [52], Ferraro and Ellis [53], and Rand and Hoen [54].

The table highlights identified factors that affect nearby stakeholders' attitudes towards nearby wind farms. These studies show that prior experience of wind farm developments can, but does not necessarily, enhance acceptance among local residents. The findings indicate that acceptance can increase over time if the development and operation of a wind farm is felt to be necessary and is consistent with stakeholders' expectations for distributive and procedural fairness [29,22,51,55]. Research into these effects temporally, proximally, and in conjunction with community/individual circumstances are discussed next in turn.

# 2.1. Temporal experiences of wind energy

In a landmark paper, Wolsink [9] observes that the acceptance of a wind farm by the community develops over the phases of a wind farm's development. The process is illustrated as following a U-shaped trajectory which describes that the first announcement of a wind farm proposal prompts residents to interrogate the acceptability of a development which can weaken supportiveness during planning, however acceptance can recover over time once the wind farm becomes operational [9]. Wilson and Dyke [23] perform a similar investigation at Roskrow Barton wind farm at Cornwall, United Kingdom, to contrast post-construction opinions towards the 'controversial' project with recalled attitudes before construction. It concludes that residents' attitudes reflect complex individual 'acceptance curves' based on specific areas of concern. While initial concerns may dissipate over the lifetime of a project, this does not imply a highly positive outlook towards the development [23].

Research suggests the negative expectations of residents are more likely to be overestimated when there is limited information about a project, but that with first-hand exposure to nearby wind farms, they may come to view the project more favourably [5,46]. This cannot be assumed however; those who are annoved by the proximity-dependent impacts of wind farms are also more likely to view the planning process as unfair to local residents [29]. A study of a Cornish wind farm finds that while residents express greater appreciation of a wind farms' visual characteristics or environmental benefits after fourteen years, there is insufficient evidence to conclude that the initially strong level of support for the wind farm varies over time [51]. This points to the complexity of societal adaptation to wind energy implementation, including throughout the post-installation processes for local involvement in the project [29,50]. For example, Mills et al. [29] and Windemer [50] note that initial perceptions of a wind farm have longer-term implications for the tolerance of wind farms throughout the operational or extended project lifetime, which is connected to stakeholders' satisfaction with the earliest processes of planning. These studies also find that the acceptance of wind energy developments reflects an evaluation of the trade-offs for local residents, and the anticipated fairness of the distribution of costs and benefits between actors involved in the project [29,50]. Similarly, a study of a Japanese wind farm development notes

that a lack of opposition towards a wind farm does not imply support for the project, and may reflect resignation [28]. Reflecting on the lack of impetus to engage local residents beyond the statutory requirements for planning consent and the associated risks for future developments, the paper calls for greater efforts to represent 'unvoiced opinions' and to inform and include 'silent' stakeholders throughout a wind farm's planning, construction and operation [28].

#### 2.2. Proximal experiences of wind energy

Evidence shows some respondents never grow accustomed to a nearby development and that proximity has at least some effect on individual attitudes [56]. Drawing on evidence collected across three suburban wind farms in Massachusetts, United States, Petrova [7] provides a framework to characterise local concerns towards specific wind farm development proposals (namely VESPA: Visual/landscape, Environmental, Socioeconomic and Procedural Aspects). Sources of local concern, such as visual impact, are often specific to the siting context with the potential to affect emotional attachments to the pre-existing character of the landscape [7,57]. Wind farm implementation therefore has the potential to disrupt the identarian values local residents ascribe to their 'backyard' [7,22,57]. NIMBY (Not in my backyard) explanations for resistance to wind farms lack compelling evidence to support that place protectionism is motivated by self-interest over the collective good [9,58,56].<sup>1</sup> Although many environmental and visual concerns about wind farms are related to proximity, residents' participation rights, access to information and trust in developers are key issues for acceptance throughout a project [7]. The study proposes a participatory 'ENUF' framework and encourages developers to 'Engage', 'Never use NIMBY', 'Understand' and 'Facilitate' local concerns [7].

Warren et al. [22] provided the first investigation into the relationship between proximity and acceptance, covering multiple projects in the early or operational phases of development across South-West Ireland and the Scottish Border. The study describes an 'inverse NIMBY' effect finding those nearest to wind farms tend to be the most supportive, concluding that their positive attitudes towards wind farms developed in the locality are engendered through positive experiences of prior wind farm implementation (p.872) [22]. The study describes resistance to wind farm developments as a NIABY (Not in anybody's backyard) rejection of wind development as a matter of principle that is reinforced by the perceived risks and/or impacts of the project for the local community [22]. Research highlights that proximity-dependent environmental impacts are a genuine source of local concern towards the siting or scale of wind farm developments. The proximity of a wind farm determines the reasonable zone of impact with a decay effect over increasing distance [59]. However, uncertainty and (mis)information about possible impacts can also reach stakeholders who live much further away [15]. Environmental arguments have been utilised by both opponents and supporters of wind energy ('green on green' debates), with the former's concerns focused on the potential impact on local flora/fauna and residents and the latter's on global environmental benefits of clean energy use [22]. Over time, this has mainstreamed the use of protective technologies and the stringency of impact assessments to mitigate repercussions and identify sources of local concern [7].

#### 2.3. Disentangling experience and acceptance

Several studies have integrated multiple factors across numerous projects to capture contextual trends in local wind energy acceptance owing to proximity and temporal exposure to wind farm developments as well as community/individual circumstances, attitudes, or involvement in renewable energy (e.g., Refs. [7,55,58,60,61]. In a choice experiment setting, Meyerhoff [60] combines latent class and spatial proximity to examine preferences for a 'constrained' or 'status quo' wind farm development programme in Germany. The findings reveal similar geographical clusters to show the opponents of wind farm proposals are more likely to live greater distances from existing wind farms and are willing to pay more on average for the siting of turbines further away [60]. Another experimental study compared Dutch citizens' attitudes and knowledge of wind energy before and after the Fukushima nuclear accident of 2011 and applied a latent class model to examine preference heterogeneity towards renewable energy development programmes. Despite the rapid deceleration of nuclear energy observed in the wake of the accident, the study suggests that individual preferences remain temporally stable, with changes observed within specific segments rather than on aggregate [61]. Hoen et al. [55] is an example of a national study with sampling based on geographic criteria and provides evidence of two parallel phenomena in a survey of United States households within 8 km of a wind farm. First, the paper proposes that attitudes which become more positive over the long-term operation of a wind farm can be partly attributed to Tiebout sorting [55].<sup>2</sup> Second, controlling for demographic differences and reported visual/auditory impact, the paper also reports that residents closer to wind turbines generally have more positive attitudes towards wind energy infrastructure [55].

Noting these potential social costs, policy increasingly highlights the importance of perceived fairness in the outcomes of wind energy development. Within this context, financial benefit-sharing mechanisms have gained prominence and tend to focus on residents who live closest to a wind farm [7,23,58]. Incentives can take the form of community benefit funds for spending on collective initiatives, structured compensation schemes, or in-kind funding from developers for projects such as infrastructure upgrades, which can provide genuine local benefit [28,29, 51]. However, research also shows that compensation has a shorter-term impact on the acceptance of landowners than local involvement and representation in the development process [29], and furthermore that the strength of the compensation effect is halved when comparing payments towards households not hosting a wind farm on their property with those that do [55]. In some instances, analysts report that the introduction of financial incentives can contribute to conflict within the community depending on who stands to gain from the operation of a wind farm [29,58].

Research into the relationship between experience and the local acceptance of wind farms presents an argument to encourage citizen participation throughout project development, beginning before the formal process of preparing a planning application with the first stages of stakeholder consultation and public engagement during site selection [48,54]. Perceived procedural fairness has been shown to reinforce respondents' appreciation for the local benefits of the wind farms, whereas perceived unfairness during the wind farm consultation process has been linked to greater perception of adverse impacts [29,44,45]. Involving community stakeholders in wind farms also serves a normative rationale to address local concerns and support justice throughout the planning, construction, and operation of a wind farm [5]. Inclusive processes imply a default position which is more conducive to local acceptance than so-called 'decide-announce-defend' siting approaches [9,58]. Communication and information-sharing can help to recognise the underlying institutional and social grievances of a silent majority [28]. Furthermore, stakeholder engagement opens alternative routes for citizens to influence decision-making rather than through legal opposition, which is said to represent the interests of a vocal but active and

<sup>&</sup>lt;sup>1</sup> NIMBY explanations for local resistance towards energy infrastructure grew from social challenges associated with nuclear facility siting in the 1970s–1980s. Research suggests its use in the context of renewable energy development is inappropriate [57,107,130].

 $<sup>^2\,</sup>$  Tiebout sorting describes the self-selection of individuals supportive of wind energy into an area hosting wind turbines, and the movement of non-supporters out of the area.

# organised minority [51].

Residents' concerns and priorities for wind farm developments determine under which conditions, and to what extent, they perceive a wind farm proposal in the community to be fair. This study investigates residents' willingness to accept wind farms within a choice experiment based on conditions which have been identified as important for distributive and procedural fairness [62]. The study aims to characterise to what extent the phase of a wind farm's development and/or proximity affects residents' preferences, expectations and acceptance.

# 3. Methods

#### 3.1. Sample and survey structure

There are approximately 2200 turbines across 300 wind farms representing 4.3 GW onshore wind capacity in Ireland distributed as shown in Fig. 1. In Ireland, almost all wind farms are owned and operated by private or semi-state companies [63,64].<sup>3</sup> Onshore renewable energy auctions aim to add 14,000 GWh by 2025 compared with 10,729 GWh in 2020, and the first and second rounds in 2020 and 2022 respectively added a further 33 wind farms, amounting to 893 MW [65,66].

Against this background, a novel survey instrument was designed to investigate wind farm neighbours' concerns for further onshore wind development. Data were collected through an online survey conducted in two stages over May–July 2021 and July–August 2022. The survey was administered by a market research company. It focused on residents living near to existing wind farms at the planning, construction, or



Fig. 1. Distribution of operating (blue) and planned (red) wind turbines (Data courtesy of SEAI, 2022).

operation phase and asked respondents about their attitudes towards another (new) wind farm located near to their home. The survey screened respondents on the basis of their age, gender, and region to identify a suitable sample living with 10 km of a wind farm as summarised in Table 2 (n = 1109). This radius defines an area of interest that is in keeping with earlier surveys of wind farm neighbours [55,22].

The survey instrument was structured into sections to elicit information from respondents to examine each of the variables shown in Table 3. The questions were designed to capture information on a fourpoint Likert scale as to respondents' trust in information sources regarding wind farms and their attitudes towards onshore wind energy as a national electricity resource. The survey instrument collected demographic information about respondents, their households, communities, and nearby wind farms. Respondents were asked to provide information about the distance to their nearest wind farm and its phase of project development at the time of the survey.<sup>4</sup> The survey also contained a choice experiment that modelled respondents' willingness to accept a hypothetical wind farm proposed for development in their community.

The choice experiment method elicits stated preference information from respondents. The method is increasingly applied to policy research [73–76] to investigate complex environmental preferences in instances where there is a lack of information concerning choice behaviours in the real market setting [77], as in the case of wind energy [74,35,78]. Choice experiments rely on the principle that preferences are based on the specific attributes of goods [79]. Using this information to elicit trade-offs between different attributes in combination presents respondents with many opportunities to express their environmental preferences, resulting in deep sampling of preferences across choice tasks [77,80].

In this study, this was achieved by defining a set of key attributes and

Table 2Demographic characteristics of the sample.

|               | Within 10 km of a wind farm (n=1109) | National census <sup>a</sup> | Difference |
|---------------|--------------------------------------|------------------------------|------------|
| Gender        |                                      |                              |            |
| Male          | 44 %                                 | 49 %                         | (-5 %)     |
| Female        | 56 %                                 | 51 %                         | (+5 %)     |
| Age           |                                      |                              |            |
| 18–24         | 20 %                                 | 11 %                         | (+9 %)     |
| 25–34         | 27 %                                 | 18 %                         | (+9 %)     |
| 35–44         | 19 %                                 | 21 %                         | (+2 %)     |
| 45–54         | 14 %                                 | 18 %                         | (-4 %)     |
| 55–64         | 10 %                                 | 14 %                         | (-4 %)     |
| 65+           | 10 %                                 | 18 %                         | (-8 %)     |
| NUTS 3 Region | 5                                    |                              |            |
| Border        | 11 %                                 | 8 %                          | (+3 %)     |
| Dublin        | 18 %                                 | 28 %                         | (-10 %)    |
| Mid-East      | 11 %                                 | 15 %                         | (-4 %)     |
| Midlands      | 7 %                                  | 6 %                          | (+1 %)     |
| Mid-West      | 13 %                                 | 10 %                         | (+3 %)     |
| South-East    | 11 %                                 | 9 %                          | (+2 %)     |
| South-West    | 18 %                                 | 14 %                         | (+4 %)     |
| West          | 10 %                                 | 10 %                         | (0 %)      |

<sup>a</sup> Statistics reported as per 2022 national census data available from the Central Statistics Office, cso.ie. Values may not sum to 100 % due to rounding error.

<sup>b</sup> Nomenclature of territorial units for statistics. Dublin intentionally undersampled as there are only a few turbines within 10 km of the city.

 $<sup>^3</sup>$  The details of connected and contracted wind farms can be viewed at the Irish Wind Atlas, gis.seai.ie/wind/.

<sup>&</sup>lt;sup>4</sup> Distances to nearest wind farms were also examined at the level of the postal routing key system using geographic coordinates. Straight-line responses were screened out of the sample.

# Table 3 Questions presented to respondents to examine variable effects.

 $\overline{\phantom{a}}$ 

| Variable  | Question as presented to respondents  | Anticipated effect on acceptance                                      | Relevant research  |
|---|---|---|--|
| Experience:   |   |   |  |
| Distance to the nearest wind farm   | (1) within 2 km, (2) 2–5 km, (3) 5–10 km  | Increasing with proximity   | Hoen <i>et al.</i> [55]<br>Warren <i>et al.</i> [22]   |
| Phase of wind farm development  | (1) Planning, (2) Construction, (3) Operation   | Increasing post-construction  | Wolsink [9]<br>Hoen <i>et al.</i> [55]   |
| Awareness of renewable energy initiatives (agree/disagree)  | <ol> <li>Is there a Sustainable Energy Community (SEC) in your area?</li> <li>Has your household ever availed of a SEAI Home Energy grant?</li> <li>Do you have investments/shares in wind energy?</li> <li>Would you be willing to get involved in a community wind farm development?</li> <li>I would like to purchase exclusively renewable 'green' electricity.</li> </ol>  | Increasing with prior involvement                                     | Bauwens and Devine-Wright [14]<br>Sirr et al. [67]   |
| We are interested in your opinion on wind energy as an electricity<br>resource<br>for Ireland. Please indicate whether you agree or disagree with the<br>following statements (agree/disagree/don't know) | <ol> <li>Wind energy in Ireland is a clean renewable energy source.</li> <li>Wind energy is key to achieving Ireland's carbon reduction commitments.</li> <li>Unreliable because the wind does not always blow.</li> <li>Does more environmental harm than good.</li> <li>Wind farm developments can negatively impact tourism.</li> <li>There is a role for both onshore and offshore wind energy.</li> <li>Wind farm developments should be moved offshore.</li> <li>Wind energy has the potential to create jobs.</li> <li>Wind farms bring discord into communities.</li> </ol> | Increasing with agreement with perceived environmental/local benefits | Devine-Wright and Howes [57]<br>Growth and Vogt [68]<br>Mills <i>et al.</i> [29]<br>Petrova [7]<br>Slattery <i>et al.</i> [69]<br>Swofford and Slattery [56] |
| How likely are you to trust the following sources of information<br>regarding the proposed wind farm? (very unlikely/unlikely/likely/<br>very likely)   | <ol> <li>A website run by the developer</li> <li>A website run by an anti-wind farm segment</li> <li>Local paper</li> <li>Local radio</li> <li>An elected representative such as a councillor or TD</li> <li>Social media such as Twitter and Facebook</li> <li>Community liaison officer from the wind farm developer</li> <li>An information pack from the wind farm developer</li> <li>An eighbour who probably knows more about these things</li> </ol>   | Increasing with greater trust in developer-supplied information       | Borch <i>et al.</i> [70]<br>Devine-Wright and Howes [57]<br>Firestone <i>et al.</i> [71]<br>Motosu and Maruyama [28]<br>Petrova [7]                          |
| Controlling variables   | (1) Gender, (2) Home ownership, (3) Land ownership, (4) Age, (5) Education, (6) Household income, (7) Tenure within the community, (8) Rural/urban area, (9) Community spirit   | Contextual preferences moderate willingness to accept                 | Hoen <i>et al.</i> [55]<br>Hyland and Bertsch [72]<br>Roddis <i>et al.</i> [40]<br>Devine-Wright and Howes [57]  |

#### Table 4

Attributes and levels included in the choice experiment.

[14]

J. le

Maitre et al.

| Attribute  | Description as presented to respondents  | Levels (trade-offs)  | Relevant research   |
|--|--|--|---|
| Distance to the wind farm<br>Visual impact               | Distance of the wind farm to your house<br>When "yes" imagine you will see one or more turbines from your house. When "no" imagine you cannot see wind<br>turbines from your house                 | <ol> <li>Outside immediate community (3–10 km)</li> <li>Close to community (2–3 km)</li> <li>Close to community (1–2 km)</li> <li>Near neighbour (within 1 km radius)</li> <li>Yes</li> <li>No</li> </ol>  | Bauwens and Devine-Wright [14]<br>Brennan et al. [33]<br>Hoen et al. [55]<br>Peri et al. [59,84]<br>Devine-Wright [85]<br>Hallan and Conziler [1] |
| Engagement during site selection                         | Modes through which the developer informs you and your neighbours of their desire to build the wind farm   | <ul> <li>(1) One-on-one with a community liaison officer</li> <li>(2) Meeting of a small segment of people living very close to the proposed site</li> <li>(3) Town hall style meeting that includes the wider community</li> <li>(4) Receive a leaflet from the developer</li> <li>(5) The developer uses local radio and newspaper</li> </ul>                                    | Strazzera <i>et al.</i> [74]<br>Bidwell [86]<br>Langer <i>et al.</i> [87]<br>Walker and Baxter [88]   |
| Community liaison officer <sup>a</sup>                   | A community liaison officer is an available point of contact for locals to raise any problems they are experiencing due to construction  | to announce their plans<br>(1) Yes<br>(2) No   | Brennan <i>et al.</i> [33]<br>Dywer and Bidwell [83]<br>Firestone <i>et al.</i> [71]  |
| Governance of the<br>community benefit fund <sup>b</sup> | The community benefit fund must be managed in a fair and transparent manner, and proposals must be assessed for their value to the community, fairness and sustainability                          | <ol> <li>Local authority</li> <li>Wind farm developer with community input</li> <li>Local enterprise development company</li> <li>Local development company with community input</li> <li>Committee compared of community members</li> </ol>   | Bristow et al. [89]<br>Devine-Wright and<br>Sherry-Brennan [90]<br>Kerr <i>et al.</i> [82]  |
| Sharing of the community<br>benefit fund                 | Sharing of the $\&$ 180,000 annual community benefit fund between community projects for the common good and personal payments to neighbours within 2 km   | <ul> <li>(1) €130,000 p.a. to community projects and</li> <li>(50,000 p.a. shared between neighbours</li> <li>(2) €100,000 p.a. to community projects and</li> <li>€80,000 p.a. shared between neighbours</li> <li>(3) €80,000 p.a. to community projects and</li> <li>€100,000 p.a. of between pairshours</li> </ul>  | Brennan et al. [33]<br>García et al. [91]<br>Vuichard et al. [92]   |
| Opportunity for citizens to<br>invest <sup>c</sup>       | An opportunity for citizens to invest in the development once planning and grid connection are granted   | <ul> <li>(1) Citizens must live close to the windfarm (within a radius of 10 km)</li> <li>(2) Citizens can live anywhere in Ireland</li> <li>(3) There is no opportunity to invest</li> </ul>  | Curtin et al. [93]<br>Sirr et al. [67]<br>Lienhoop et al. [39]<br>Jørgensen et al. [94]<br>Walker and Baxter [88]                                 |
| Developer <sup>d</sup>                                   | The wind farm developer offering the investment (Commercial wind farms can produce electricity at a lower cost than either a community-led development or a joint developer/community development) | Community ownership (30 %/50 % extra cost):<br>(1) Local landowners/farmers<br>(2) A community sustainable energy<br>(3) A new venture for an existing community<br>network group<br>Community (co-)ownership (20 % extra cost):<br>(4) Joint developer and community venture<br>Private/state ownership:<br>(5) Irish semi-state companies<br>(6) Private companies or developers | Baxter <i>et al.</i> [25]<br>Goedkoop and Devine-Wright<br>[95]<br>Philpott and Windemer [96]<br>Slee, [97]<br>Warren and McFadyen [98]           |

<sup>a</sup> The Code of Practice for Wind Energy Development in Ireland provides guidelines for community engagement including the appointment of a community liaison officer, which is required for new wind farms developed after 2020 [99].

<sup>b</sup> The Irish national government recommends that the administration of community benefit funding is carried out by a committee of five to fourteen people including the developer and an administrator, with decisionmaking carried out by a panel of community representatives [81].

<sup>c</sup> There is discussion about mechanisms to encourage Irish citizens to invest into onshore wind farms. The first auction rounds included preferential categories for community-led renewable energy projects [99].

<sup>d</sup> Ireland's first two onshore renewable energy auctions contain a preferential capacity category for community bids, provided at least one member is part of a Sustainable Energy Community registered with the state agency for renewable energy [99]. Based on the average strike-price of successful projects awarded under the first onshore renewable energy auction [65], the *Developer* attribute was coupled to an attribute describing the *cost of electricity production* as shown. Levels of increase (20 %, 30 %, 50 %) were designed in consultation with policymakers and industry. Incompatible levels were not presented in combination.

Table 5

Goodness-of-fit measures for latent class solutions.

| Segments | Log-likelihood | Percent Certainty <sup>a</sup> | Chi-Square | Relative Chi-Square <sup>b</sup> | AIC      | CAIC <sup>c</sup> | BIC <sup>c</sup> |
|----------|----------------|--------------------------------|------------|----------------------------------|----------|-------------------|------------------|
| 1        | -10975.08      | 3.40                           | 772.02     | 29.69                            | 22002.15 | 22218.46          | 22192.46         |
| 2        | -9610.73       | 15.41                          | 3500.71    | 66.05                            | 19327.46 | 19768.40          | 19715.40         |
| 3        | -9315.59       | 18.00                          | 4090.99    | 51.14                            | 18791.19 | 19456.75          | 19376.75         |
| 4        | -9248.90       | 18.59                          | 4224.37    | 39.48                            | 18711.80 | 19601.99          | 19494.99         |
| 5        | -9208.46       | 18.95                          | 4305.25    | 32.13                            | 18684.92 | 19799.74          | 19665.74         |
|          |                |                                |            |                                  |          |                   |                  |

<sup>a</sup> Percent certainty produces a goodness-of-fit measure analogous McFadden's pseudo-R<sup>2</sup> by using the geometric mean of the model likelihood.

<sup>b</sup> Relative Chi-Square applies a penalty for each added segment, and is twice the log-likelihood of the current model and twice the log-likelihood of the null model, divided by the number of parameters.

<sup>c</sup> Consistent Akaike Information Criterion is a common method to determine the number of segments and is similar to the Bayesian Information Criterion (BIC) by adjusting the log likelihood for the number of classes, choice tasks and independent parameters estimated for each class.

levels for the local acceptance of a wind farm at the planning, construction and operation phases of development based on the existing research as shown in Table 4 [62]. The attributes included in the analvsis are based on existing mechanisms aimed at promoting distributive and procedural justice at different phases of wind farm development [62]. For example, local financial benefits associated with wind farm developments are increasingly formalised in Ireland [81] and are generally paid on an annual basis by professional developers into a community benefit fund or trust shared between, and governed by, residents living within the locality [82,50]. Another approach recommends that developers employ a community liaison officer to act as a single point of contact with the community, with the aim of better mediating and addressing residents' concerns about the wind farm [83, 62]. The levels of each attribute were used to define a set of individual-specific wind farm proposals comprising one level of each attribute in a randomised fashion. Maintaining even sampling balance, respondents were presented with a set of choice tasks representing a proposal for the development of a wind farm within the community. This required respondents to make a series of eight trade-offs between two different wind farm proposals. Willingness to accept each development was incorporated into the choice experiment using a dual-response task that appeared after each choice to ask respondents whether they would accept their preferred wind farm if it were proposed for development within their community.<sup>5</sup> The sampling depth over attributes associated with each phase of wind farm development allows for measurement relative to respondents' willingness to accept. This is modelled against the dual-response task which represents preferences for the status quo.

#### 3.2. Latent class segmentation

Stemming from consumer demand theory [79] and utility maximisation theory [100], the choice experiment methodology assumes that with some degree of natural variability, individuals (*i*) will consider wind farm proposals based on their defining characteristics (*j*) and choose the proposal which provides the greatest personal satisfaction (economic utility). The indirect total utility associated with wind farm development ( $U_{ij}$ ) can be expressed under a model with a vector of estimable part-worth utility coefficients ( $\beta_j$ ) describing the wind farm attributes ( $X_{ij}$ ), such that the probability function Pr(ij), is given by:

$$Pr(ij) = \frac{exp(U_{ij})}{\sum exp(U_{ik})} = \frac{exp(\beta_j X_{ij})}{\sum exp(\beta_j X_{ik})} \forall j \neq k$$
(1)

It is advantageous to understand why individuals' utilities share similarities, however this often depends on unobserved characteristics which are difficult to predict *a priori* [73]. Following similar approaches

[26,60,74–76], this paper uses a latent class utility model to assign respondents with similar willingness to accept to segments according to their behaviour within the choice experiment. This assumes the probability of respondent choices depends on  $h_{LC}$ , the prior probability of latent class membership, and Pr(ij|LC), the conditional probability of a choice given that the class membership effects of individuals with similar latent preferences, expressed as follows [74,77]:

$$Pr(ij) = \sum h_{LC} Pr(ij|LC) \tag{2}$$

Following Tabi et al. [75], the latent class model defines segment membership and part-worth utility estimates. Part worth utilities,  $\beta_{illC}$ , are calculated at the individual-specific level using hierarchical Bayes estimation.<sup>b</sup> The aggregate model obtained a good fit with a percent certainty of 0.603 and root likelihood of 0.668. The latent class procedure was replicated five times from random starting points for up to five-class solutions under Sawtooth Software's default settings. Table 5 indicates that the three-class model was selected as the most parsimonious model, given trade-offs between goodness-of-fit measures (relative chi-square and percent certainty) and over-parameterisation as detected by the Consistent Akaike Info Criterion [75]. Maximum average segment membership is high, 93.7 %. The first latent class contains 397 respondents (35 %) who in aggregate accepted 66.3 % (SE = 8.4 %) of the wind farm proposals they encountered within the choice experiment. The second latent class contains 144 respondents (13 %) who accepted just 4.4 % (SE = 0.7 %) of wind farm proposals within the choice experiment. The third latent class contains 587 respondents (52%) who accepted almost every proposal within the choice experiment (99.3 %, SE = 1.1 %).

# 3.3. Analysis and profiling

Willingness to accept was investigated using a randomised firstchoice method. Randomised first-choice considers trade-offs between each level of each attribute and extends the simple total utility methods and simulates naturally imperfect choice behaviour by incorporating variability into the error term [73,101]. It predicts the share of preference between the status quo (i.e., no further wind development) and the

<sup>&</sup>lt;sup>5</sup> The dual-response option allowed respondents to indicate after each choice whether they would take this option if it was presented to them in reality.

<sup>&</sup>lt;sup>6</sup> Hierarchical Bayes regression is a multilevel approach that advantageously separates meaningful differences in individual preferences from random variance. The estimation method iteratively converges on an individual's specific utility function by borrowing information from other respondents. It is advantageous to combine the strengths of the latent class procedure, (i.e., to identify discrete differences in preferences between segments), with hierarchical Bayes estimation, (i.e., to converge on the mean part-worth utility for each segment to provide individual-specific parameters free from the independence of irrelevant alternatives (IIA) assumption) [100].

wind farm development conditions characterised in Table 2 [100]. A preferred 'best-case' combination of wind farm characteristics was included which maximises respondent utility.<sup>7</sup> Willingness to accept was compared across all attribute levels, *ceteris paribus*. This examined the influence of specific development conditions on acceptance including the best-case combination as well as the option not to accept a wind farm development under any circumstances. Latent classes' preferences between attributes were expressed as a relative importance measure captured by the range of individual-specific utilities for an attribute of interest over all attributes.

To profile each latent class, the factors affecting willingness to accept further wind energy development in the community, were profiled under a multinomial probit approach.<sup>8</sup> The results were checked for robustness.<sup>9</sup> Table 6 contains the descriptive statistics of each variable (experience over project phase/proximity; prior un/involvement in renewable energy; attitudes towards onshore wind electricity; mis/trust in information sources) across the full sample and for each latent class. Variables with multiple levels each sum to 100 % by attribute down each column, notwithstanding rounding error. Corresponding chi-square goodness-of-fit measures indicate that there are significant bivariate differences between the three latent classes.<sup>10</sup>

#### 4. Results and discussion

This section investigates whether residents living near to a wind farm have similar attitudes towards additional wind energy developments proposed in the community based on their prior experience and the conditions of the proposal. First, Fig. 2 illustrates specific trends in willingness to accept and adds to existing research examining prior experience [2,55,102] over the phase of development and the proximity to the wind farm. Second, Fig. 3 and Table 7 summarise the wind farm preferences of three latent class segments identified using individual-specific information collected within the choice experiment. These are applied in Table 8 which profiles each segment to identify the key considerations for acceptance.

#### 4.1. Proximity, phase of development and acceptance

This section investigates the specific effects of prior experience owing to proximal or temporal exposure to existing wind farms on residents' willingness to accept a new wind farm proposed for development in the community. Fig. 2 compares the share of preferences for a 'bestcase' wind farm proposal which is predicted to maximise utility with

$$\begin{split} & \Sigma \beta_3(\text{Controls}_i) + \varepsilon_i \ \text{Full model } 2: \text{LC}_i \ = \ \beta_0 + \ \beta_1(\text{Phase}_i) + \ \beta_2(\text{Proximity}_i) + \\ & \Sigma \beta_3(\text{Controls}_i) + \ \Sigma \beta_4(\text{Attitudes}_i) + \ \Sigma \beta_5(\text{Information}_i) + \varepsilon_i. \end{split}$$

preferences for the status quo. The figure contrasts residents living near a wind farm at the planning, construction or operation phase of development. It also compares between those living 2 km, 2–5 km, and 5–10 km away from an existing wind farm.<sup>11</sup>

The results of Fig. 2 highlight that considering the proximity between residential areas and wind farms (within 2 km) and the earliest phases of a wind farm's development is critical for acceptance. Specifically, Fig. 2A shows in light bars that on average willingness to accept wind farm developments in the community is stronger for residents at the construction or operation phase of development than in the planning phase. This trend is significant at the p < 0.01 level as indicated by asterisks. Similarly, the dark bars in Fig. 2B show that residents living 2–10 km from a wind turbine have similar willingness to accept. However, respondents living within 2 km are less willing to accept further wind farm development compared to those living further away (p < 0.05). The grey bars indicate that more residents at the planning phase and within 2 km are likely to prefer maintaining the status quo and forgo further wind farm development than the rest of the sample.

Earlier research has proposed that the significant shift in attitudes at the earliest phase of planning are a function of the responsive interest to the announcement of a wind farm and speculation about the possible future impacts of such a development [9]. The uncertainty of a long-term proposal requires communities to collectively and separately interpret, evaluate, and act on information communicated through local media and the developer [57,85]. This suggests uncertainty is likely to be strongest for individuals within the immediate vicinity of a wind farm [84]. Other papers identify a stronger relationship between proximity and acceptance, whether reinforcing [e.g., Ref. [55]] or counteractive [e.g., Ref. [33]]. Elsewhere, spatial analysis reports a non-linear trend, remarking that regions with the greatest concentration of onshore energy infrastructure (e.g., rural Scotland) and also the least concentration (e.g., city of London) show comparatively low levels of support for further energy development [40].

In order to examine the relationships between acceptance and residents' experience owing to the proximity and phase of existing wind farms, a latent class and profiling procedure was performed as detailed in Subsections 4.2 and 4.3 respectively.

# 4.2. Latent class preferences

A latent class model for residents' preferences for additional wind farm developments in their community identified three segments whose level of supportiveness for additional implementation showed clear distinctions (unwillingness to accept; strong willingness to accept; and a segment expressing moderate willingness to accept). The choices made by each segment indicate the preferred conditions for a wind farm development in the community. Table 7 and Fig. 3 highlight key differences between the preferences of each segment. Table 7 reports the relative importance of attributes. This provides an overview of the features which most strongly influenced respondents' choices. Fig. 3 compares respondents' preferences based on the conditions of each wind farm proposal. It represents willingness to accept as a share of preference comparing each condition against the average 'best-case' wind farm and the option to refuse further wind development (i.e., opt out).<sup>12</sup>

Approximately 13 % of respondents are unwilling to accept wind farm developments under any circumstances. Unwillingness to accept wind farm development is reflected as a near-zero share of preference

 $<sup>^7</sup>$  Individual-specific part-worth utilities are provided in Appendix Table A1. From these, a randomised first choice procedure predicts on average 79.8 % (SE = 0.8 %) would be willing to accept such a wind farm proposal.

<sup>&</sup>lt;sup>8</sup> Multinomial probit regression suitably accommodates random preference heterogeneity, substitution, and error correlation. Dummy variables test the hypotheses that willingness to accept further wind development within the community is enhanced by experience owing to the phase a wind farm development ( $\beta_1$ ), proximity to a wind farm ( $\beta_2$ ), controls for sociodemographic context and awareness of renewable energy initiatives ( $\beta_3$ ), un/certainty as to the role of onshore wind energy as a source of national electricity ( $\beta_4$ ), and mis/trust in information sources concerning wind developments ( $\beta_5$ ) (Table 3). Constrained model 1 : LC<sub>i</sub> =  $\beta_0 + \beta_1$ (Phase<sub>i</sub>) +  $\beta_2$ (Proximity<sub>i</sub>) +

<sup>&</sup>lt;sup>9</sup> The interaction term (Phase)\*(Proximity) was included as an additional regressor in models 1 and 2 (see Footnote 7) to assess whether the joint temporal and spatial effect has a significant influence on segment membership. It was found to be insignificant. The significance, magnitude, and direction of other coefficients in models 1 and 2 did not change. Results are available from the authors.

 $<sup>^{10}</sup>$  Correlation matrices for each variable are provided in the Appendix, Fig. A1.

<sup>&</sup>lt;sup>11</sup> Further contrasts between the average frequency of acceptance over time for wind farm neighbours living within the immediate radius of a wind farm (within 2 km) or further away (up to 10 km) are provided in the Appendix. This approximates the underlying propensity of each segment of wind farm neighbours to accept a wind farm.

<sup>&</sup>lt;sup>12</sup> The 'best-case' wind farm uses the individual-specific total utility. Individual-specific utilities are provided in the Appendix, Table 1.

# Table 6

Descriptive statistics for the full sample and each latent class.

|  | Unwillingness to accept (13 %) | Moderate willingness to accept (35 %) | Strong willingness to accept (52 %) | Total (100 %) | p-value <sup>a</sup> |
|--|--------------------------------|---------------------------------------|-------------------------------------|---------------|----------------------|
| Wind farm characteristics                                  |                                |                                       |                                     |               |                      |
| Distance to the wind farm                                  |                                |                                       |                                     |               | 0.001                |
| Within 2 km  | 28 %                           | 15 %                                  | 14 %                                | 16 %          |                      |
| 2–5 km   | 33 %                           | 38 %                                  | 42 %                                | 39 %          |                      |
| 5–10 km  | 39 %                           | 47 %                                  | 45 %                                | 45 %          |                      |
| Phase of wind farm development                             |                                |                                       |                                     |               | 0.000                |
| Planning   | 35 %                           | 27 %                                  | 15 %                                | 22 %          |                      |
| Construction   | 6 %                            | 15 %                                  | 9%                                  | 11 %          |                      |
| Operation  | 60 %                           | 58 %                                  | 76 %                                | 68 %          |                      |
| Individual characteristics                                 | (D. 0/                         | 50.0/                                 | 50.0/                               | FC 0/         | 0.005                |
| Gender (Female)  | 62 %                           | 59 %                                  | 52 %                                | 56 %          | 0.035                |
| Homeowner  | 87 %                           | 70 %                                  | 75 %                                | 74 %          | 0.000                |
| Landowner  | 38 %                           | 38 %                                  | 38 %                                | 38 %          | 0.986                |
| Age  | 10.0/                          |                                       | 40.0/                               | 47 0/         | 0.000                |
| 18-34  | 19 %                           | 55 %<br>21 0/                         | 49 %                                | 47 %          |                      |
| 55-54<br>FE -  | 38 %<br>42 %                   | 31 %                                  | 33 %<br>19 0/                       | 33 %<br>20 %  |                      |
| 55+  | 42 %                           | 14 %                                  | 18 %                                | 20 %          | 0.1.40               |
| Education  | 7.0/                           | 11.0/                                 | 14.0/                               | 10.0/         | 0.149                |
| Primary level, junior cycle (NFQ 1–3)                      | /%                             | 11 %                                  | 14 %                                | 12 %          |                      |
| School leaving certificate (NFQ 4–5)                       | 24 %                           | 25 %                                  | 25 %                                | 25 %          |                      |
| Third level and higher education (NFQ 6–10)                | 69 %                           | 64 %                                  | 61 %                                | 63 %          | 0.000                |
| Employment status  | 49.0/                          |                                       | E7 0/                               |               | 0.000                |
| Working rull-time  | 40 %                           | 55 %<br>20 %                          | 5/%<br>100/                         | 55 %          |                      |
| working part-time  | 10 %                           | ∠U %                                  | 18 %                                | 1/ %          |                      |
| Not currently working                                      | 24 %                           | 20 %                                  | 16 %                                | 18 %          |                      |
| Retired  | 18 %                           | 6 %                                   | 10 %                                | 9%            |                      |
| Household characteristics                                  |                                |                                       |                                     |               | 0.155                |
| Annual household income                                    | 05.04                          | 00.0/                                 | 01.0/                               | 00.0/         | 0.177                |
| t23,348 or under   | 25 %                           | 23 %                                  | 21 %                                | 22 %          |                      |
| £23,400 - 62,348   | 51 %                           | 45 %                                  | 44 %                                | 45 %          |                      |
| t62,400 or over  | 24 %                           | 32 %                                  | 35 %                                | 32 %          | 0.001                |
| Number of people in household                              | <b>24</b> 44                   |                                       | 10.00                               | 10.00         | 0.001                |
| 1 person   | 21 %                           | 11 %                                  | 12 %                                | 13 %          |                      |
| 2 people   | 43 %                           | 44 %                                  | 37 %                                | 40 %          |                      |
| 3+ people  | 36 %                           | 45 %                                  | 51 %                                | 47 %          |                      |
| Community characteristics                                  |                                |                                       |                                     |               | 0.000                |
| lime living in current community                           | 2.0/                           | <i>C</i> 0/                           | 0.0/                                | 4.0/          | 0.000                |
| Under I year   | 3 %                            | б %<br>07.0/                          | 3 %                                 | 4%            |                      |
| 1–10 years   | 20 %                           | 37 %                                  | 40 %                                | 37 %          |                      |
| 11–20 years  | 21 %                           | 24 %                                  | 20 %                                | 22 %          |                      |
| Over 20 years  | 56 %                           | 33 %                                  | 36 %                                | 38 %          | 0.000                |
| Rural Area (nonvelation under 1500)                        | 71.0/                          | 40.0/                                 | 40.0/                               | 44.0/         | 0.000                |
| Rural Area (population under 1500)                         | /1 %                           | 40 %                                  | 40 %                                | 44 %          |                      |
| Sinail town (population between 1500 and 5000)             | 17 %                           | 27 %                                  | 20 %                                | 25 %          |                      |
| A site (nonviolation groater than 50,000)                  | 9 %                            | 30 %                                  | 20 %                                | 25 %          |                      |
| A city (population greater than 50,000)                    | 3 %                            | 3 %                                   | 7 %0                                | 5 %           |                      |
| Awareness of renewable energy initiatives (yes, %)         | 0.04                           | 20.0/                                 | 06.04                               | 20.0/         | 0.000                |
| Know of a sustainable energy community in the area         | 9%                             | 30 %                                  | 36 %                                | 30 %          | 0.000                |
| Availed of a nome energy grant                             | 19 %                           | 25 %                                  | 31 %                                | 28 %          | 0.008                |
| Willing to get involved in a community wind form           | 3 %0<br>9 0/                   | 18 %                                  | 20 %                                | 17 %          | 0.000                |
| While to get involved in a community wind farm             | 8 %                            | 29 %                                  | 33 %<br>60 %                        | 29 %          | 0.000                |
| Strange community primiting the community                  | 49 %                           | 60 %                                  | 69 %                                | 63 %          | 0.000                |
| Strong community spirit in the community                   | 81 %                           | 86 %                                  | 90 %                                | 87 %          | 0.023                |
| Opinions concerning wind electricity (agree, %)            | (D. )/                         | 00.0/                                 | 00.0/                               | 06.04         | 0.000                |
| A clean renewable energy source                            | 62 %                           | 89 %                                  | 89 %                                | 86 %          | 0.000                |
| Key to achieving carbon reduction commitments              | 42 %                           | // %                                  | 83 %                                | /5 %          | 0.000                |
| Unreliable because the wind does not always blow           | 38 %                           | 30 %                                  | 29 %                                | 31 %          | 0.094                |
| Does more environmental harm than good                     | 40 %                           | 20 %                                  | 21 %                                | 23 %          | 0.000                |
| Wind farm developments can negatively impact tourism       | 65 %                           | 37 %                                  | 25 %                                | 34 %          | 0.000                |
| I nere is a role for both onshore and offshore wind energy | 4/ %                           | /8 %                                  | 85 %                                | 78 %<br>49 %  | 0.000                |
| Wind farm developments should be moved offshore            | 65 %                           | 44 %                                  | 37 %                                | 43 %          | 0.000                |
| while energy has the potential to create jobs              | 5U %                           | /8 %0<br>477 0/                       | 81 %                                | /0 %<br>40.0/ | 0.000                |
| while farms bring discord into communities                 | 01 %                           | 4/ %                                  | 41 %                                | 48 %          | 0.000                |
| (very) likely to trust sources of information (%):         | 20.04                          | <b>F 4</b> 0/                         | (0.0)                               | <b>F7</b> 0/  | 0.000                |
| website run by the developer                               | 2U %                           | 54 %                                  | б9 %                                | 57 %          | 0.000                |
| website run by an anti-wind farm segment                   | 42 %                           | 34 %                                  | 38 %                                | 37 %          | 0.121                |
| Local paper  | 56 %                           | 61 %                                  | /1 %                                | 65 %          | 0.000                |
| Local radio  | 55 %                           | 60 %                                  | /3 %                                | 66 %          | 0.000                |
| Elected representative such as councillor/TD               | 32 %                           | 49 %                                  | 62 %                                | 53 %          | 0.000                |
| Social media such as Twitter and Facebook                  | 18 %                           | 38 %                                  | 45 %                                | 39 %          | 0.000                |
| Community liaison officer from the wind farm developer     | 28 %                           | 58 %                                  | 76 %                                | 64 %          | 0.000                |
| Information pack provided by the developer                 | 26 %                           | 62 %                                  | 75 %                                | 64 %          | 0.000                |
| A neighbour who probably knows more about these things     | 43 %                           | 48 %                                  | 55 %                                | 51 %          | 0.013                |

Note: Categories sum down each column. For brevity 'agree', '(very) likely' and 'yes' statements are reported excluding negative/neutral responses. Due to rounding errors, totals may not add to exactly 100 %.

<sup>a</sup> Two-tailed Chi-square goodness of fit, corresponding to \*p < 0.1, \*\*p < 0.05, \*\*\*p < 0.01.

<sup>b</sup> Owned outright or with Mortgage/Loan.

<sup>c</sup> National Framework of Qualifications is a 10-level system. NFQ levels 1–5 apply to education up to the end of formal schooling. NFQ levels 6–10 apply to higher academic education, including the higher-doctoral level. Details can be viewed at nfq-qqi.com.

<sup>d</sup> Unemployed, Student, Homemaker, or Unable to work due to health reasons.

for wind development in the area, irrespective of the conditions of the project (left dotted line, Fig. 3). The relative importance measures shown in Table 7 indicate that these resident's choices are strongly influenced by the proximity and visual impact attributes and less so by the developer of the wind farm than the rest of the sample.

In contrast, 52 % of residents are strongly willing to accept additional wind farm implementation in the community. However, they perceive a higher relative importance concerning the developer of a wind farm compared to individuals who are unwilling to accept wind farm developments (Table 7, fifth column). Fig. 3 (further right curve) shows that compared to other forms of ownership, residents who are strongly willing to accept prefer wind farms owned by local sustainable energy community and local (co-)ownership opportunities, such as providing opportunities for citizens living within 10 km of a wind farm to invest into the project. This suggests that the residents who are most willing to accept wind farms in the area value opportunities for community involvement and ownership of future wind farms [25,103–105]. For these residents, engagement during site selection is also amongst the most important wind farm attributes (Table 7) and in-person community

engagement has an influence on willingness to accept across all phases of development – during siting (i.e., one-on-one, small, and town hall gatherings), construction (i.e., via a community liaison officer) and operation (i.e., administration of community benefit funding under a committee of community representatives). These identified conditions for support contribute to research which increasingly recognises the role of participatory justice in determining wind farm acceptance [25,33,9, 35,105].

A third segment containing 35 % of respondents show moderate willingness to accept (middle curve, Fig. 3; middle column, Table 7). As described elsewhere, this segment revealed a 'midpoint stance' towards further onshore wind farm development which might suggest uncertainty [14], ambivalence [61,35] or qualified support for wind development [18]. While generally individuals have similar preferences for wind development, this segment's choices reflect a greater preference for near-neighbour benefit contributions and wind farms which are further away or not visible. This finding suggests that individuals with moderate preferences take into consideration their understanding of the 'cost-benefit ratio' of wind farm proposals [21]. The results reflect the



Fig. 2. Share of preference for a 'best-case' combination of wind development features and the status quo (A) over phase of project development; and (B) over proximity to the wind farm.

Note: Non-overlap of 95 % confidence intervals indicates significance demonstrated by \*\*\* for p < 0.01. A randomised first-choice procedure was used to define a wind farm scenario based on maximised total utility.

|                           |   |      | <ul> <li>Strong willingness to accept</li> <li>Moderate willingness to accept</li> <li>Unwillingness to accept</li> </ul> | t |
|---------------------------|---|------|---|---|
|                           | Local landowners/farmers +50%   | 7    |   |   |
|                           | Community sustainable energy group +50%   | H    |   |   |
| Ownership                 | Existing community network +50%   | H    | inte  |   |
| Ownership                 | Local landowners/farmers +30%   | +    | in in   |   |
|                           | Community sustainable energy group +30%   | H    |   |   |
|                           | Existing community network +30%   | H    |   |   |
|                           | Joint developer/community venture +20%  | H    |   |   |
|                           | Irish semi state companies<br>Private companies/developers  | ET   |   |   |
|                           | Local radio and newspaper   | -    | <u>, , , , , , , , , , , , , , , , , , , </u>   |   |
|                           |   | Ŧ    | ±+,±  |   |
| -                         | Small group of near-neighbours  | H    |   |   |
| Engagement over           | r I own hall style meeting  |      |   |   |
| project phase             | One-on-one with a CLO   | H    |   |   |
|                           | CLO not available   | H    | H. H.   |   |
|                           | CLO throughout construction   | н    |   |   |
|                           | Local authority<br>Wind farm developer & community<br>Enterprise development company<br>Local develpment company &community<br>A committee of community members | II   |   | 4 |
|                           | Within 1km<br>1-2km<br>2-3km<br>3-10km  | II.  |   |   |
| Distributional<br>aspects | Visible from home<br>Not visible from home  | T    |   |   |
|                           | €50,000 towards near-neighbours   | H    |   |   |
|                           | €80,000 towards near-neighbours   | H    |   |   |
|                           | €100,000 towards near-neighbours  | Ь    |   |   |
|                           | No opportunity<br>All citizens<br>Residents within 10km   | Immi |   |   |
|                           | ~   | 0/   | 10% 20% 20% 10%   | , |
|                           | d   | 70   | 10% 20% 30% 40%   | 0 |
|                           |   |      | Snare of preference (%)   |   |

Fig. 3. Share of preference for different wind farm development conditions contrasted between three latent classes

Note: Bounded lines show 95 % confidence intervals. Non-overlap indicates significance of difference. Share of preference for attribute levels calculated by randomised first-choice versus a 'best-case' wind farm and the option for no further wind development.

#### Renewable and Sustainable Energy Reviews 189 (2024) 113839

#### Table 7

Relative importance of attributes to three latent class segments.

|                         | Attribute                          | Unwillin<br>to accep<br>%) | ngness<br>ot (13 | Moderat<br>willingn<br>accept ( | te<br>less to<br>(35 %) | Strong<br>willingn<br>accept ( | ness to<br>(52 %) | Total (100 %)  |
|-------------------------|------------------------------------|----------------------------|------------------|---------------------------------|-------------------------|--------------------------------|-------------------|--|
|                         |                                    | RI                         | SE               | RI <sup>a</sup>                 | SE <sup>b</sup>         | RI                             | SE                | 'Best-case' preferred level of the attribute                           |
| Ownership               | Developer                          | 14.2                       | 0.4              | 17.3                            | 0.3                     | 17.2                           | 0.2               | A community sustainable energy segment (30 % extra                     |
|                         |                                    | %                          | %                | %                               | %                       | %                              | %                 | electricity production cost)   |
|                         | Engagement during site selection   | 11.5                       | 0.4              | 14.3                            | 0.3                     | 15.8                           | 0.2               | One-on-one with a community liaison officer                            |
|                         |                                    | %                          | %                | %                               | %                       | %                              | %                 |  |
| Engagement over project | Community liaison officer during   | 5.3 %                      | 0.4              | 6.2 %                           | 0.2                     | 8.0 %                          | 0.2               | Yes  |
| phase                   | construction                       |                            | %                |                                 | %                       |                                | %                 |  |
|                         | Governance of the community        | 11.9                       | 0.4              | 14.1                            | 0.2                     | 13.9                           | 0.2               | A committee composed of community members                              |
|                         | benefit fund                       | %                          | %                | %                               | %                       | %                              | %                 |  |
| Distributional aspects  | Distance to the wind farm          | 27.7                       | 1.0              | 19.4                            | 0.5                     | 17.4                           | 0.4               | Over 3 km but under 10 km  |
|                         |                                    | %                          | %                | %                               | %                       | %                              | %                 |  |
|                         | Visual impact (view from home)     | 9.0 %                      | 0.5              | 7.4 %                           | 0.3                     | 6.6 %                          | 0.2               | No   |
|                         |                                    |                            | %                |                                 | %                       |                                | %                 |  |
|                         | Sharing of the community benefit   | 9.5 %                      | 0.4              | 9.7 %                           | 0.2                     | 9.8 %                          | 0.2               | If they live close to the wind farm (within 10 km)                     |
|                         | fund                               |                            | %                |                                 | %                       |                                | %                 |  |
|                         | Opportunity for citizens to invest | 10.7                       | 0.4              | 11.6                            | 0.3                     | 11.3                           | 0.2               | $\pounds$ 100,000 p.a. to community projects and $\pounds$ 80,000 p.a. |
|                         |                                    | %                          | %                | %                               | %                       | %                              | %                 | shared between neighbours  |

<sup>a</sup> Relative importance, RI (%): averaged individual utility range within each attribute as % of the total range of all attributes.

<sup>b</sup> Standard error of the mean (SE): a measure of sample heterogeneity equal to standard deviation over the square-root of sample size.

preferences of residents with some experience of existing wind farms, supporting that community benefit-sharing can help to recognise the potential local impacts of wind farms [8,23,82] and tend to be viewed as a favourable opportunity for communities in the area [33,58,82,106].

# 4.3. Factors affecting willingness to accept

This section jointly examines various factors affecting respondents' acceptance for additional wind farm development in their community. Table 8 shows the results of the multinomial probit models. Model 1 is 'constrained' to include only controlling sociodemographic variables and experience variables. Model 2 is 'full' and additionally examines the effect of attitudes towards onshore wind energy and trust in information sources as well as the sociodemographic and experience variables. The direction and significance of the relationships in comparison are of interest, holding all else constant. These are reported alongside the standard error. The Wald chi-square statistic increases between Model 1 and Model 2 showing an improvement in the predictive power with the inclusion of all variables.

First addressing each of the variables of interest, the models demonstrate that residents strongly willing to accept wind farm development in their community are significantly more likely to live within 10 km of a wind farm which is currently in operation (i.e., 76 %, Table 6) than a wind farm which is currently in its planning phase (15%). A greater proportion of residents with weaker preferences for wind farm development are at earlier phases of project development than strongly supportive respondents (Table 6). This indicates that the planning phase is likely to be an important time to address local concerns [107,53,62,45]. Furthermore, Model 1 in Table 8 also suggests that residents are more likely to be willing to accept projects that are 2-10 km away from their home rather than within a 2 km 'near-neighbour zone' of possible impact [33]. However, proximity is not a significant factor when all potentially relevant variables are considered in the full model (in Model 2). It is worth noting that the dataset reflects the Irish context, where there are more operating wind farms than in construction phases of development as well as comparatively fewer households within 2 km than 2–10 km of a wind farm. For the large majority of nearby residents, the results would suggest sited

wind farms become increasingly viewed as commonplace features over their operational lifetime [1,10,14].

Awareness of local renewable energy initiatives also has an influence on willingness to accept wind farms. Model 1 shows that respondents who are strongly willing to accept wind farm development are more likely to be aware of a sustainable energy community in the area, have a strong sense of community spirit, and express a desire to purchase exclusively 'green' electricity, unlike respondents unwilling to accept. The models provide evidence to suggest respondents moderately willing to accept are on average more willing to get involved in a communityowned wind farm initiative. Previous research has similarly described non-members of energy cooperatives as being more indifferent or uncertain towards renewable energy than members [14]. This highlights a growing understanding that acceptance is affected by local culture and social norms which influence the perceived "fit" between a wind farm development and the community [11,21,108].

Wind farm neighbours generally agree with economic and environmental arguments for wind energy, such as job creation (76 % agreement), clean energy (86 %) and carbon reduction (75 %) as reported in Table 6. Table 8 further illustrates that these viewpoints seem to have greater resonance with individuals willing to accept wind energy developments [69]. For instance, Model 2 shows that respondents who are supportive of wind electricity as a national resource are similarly likely to be willing to accept wind farms developments in the community, and vice versa. Those who are unwilling to accept wind energy are more likely to express that wind farms can cause discord in communities, offer limited local job creation, and are non-essential for reducing greenhouse gas emissions. Concerns relating to intra-community conflict, local economic benefits and environmental impact have been identified as issues for fairness that are important to both supporters and opponents of wind energy [9,58]. These findings suggest that greater sensitivity towards potential risks, rather than benefits, is associated with lower willingness to accept additional wind farms in the community. Another noticeable difference between segments concerns local tourism. Whereas residents who are strongly willing to accept appear unconvinced of the potential for negative impacts on tourism, those moderately or unwilling to accept generally agree that there could be adverse effects. These findings are consistent with what has been described as

# Table 8

15

Average marginal effects for factors affecting willingness to accept wind farm development.

|  | Model 1: 'C          | onstrained'   |                          |                      |                        |                      | Model 2: 'F          | 'ull'          |                          |                      |                            |                 |
|--|----------------------|---------------|--------------------------|----------------------|------------------------|----------------------|----------------------|----------------|--------------------------|----------------------|----------------------------|-----------------|
|  | Unwillingn<br>(13 %) | ess to accept | Moderate w<br>accept (35 | villingness to<br>%) | Strong wi<br>accept (5 | llingness to<br>2 %) | Unwillingr<br>(13 %) | ness to accept | Moderate w<br>accept (35 | villingness to<br>%) | Strong willi<br>accept (52 | ngness to<br>%) |
|  | AME <sup>a</sup>     | $SE^{b}$      | AME                      | SE                   | AME                    | SE                   | AME                  | SE             | AME                      | SE                   | AME                        | SE              |
| Distance to the wind farm (ref. within 2 km)             |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| –5 km  | -0.06*               | 0.03          | -0.02                    | 0.04                 | 0.08*                  | 0.04                 | 0.01                 | 0.02           | -0.04                    | 0.04                 | 0.03                       | 0.04            |
| -10 km   | -0.06**              | 0.03          | -0.02                    | 0.04                 | 0.08*                  | 0.04                 | 0.02                 | 0.02           | -0.03                    | 0.04                 | 0.01                       | 0.04            |
| <b>Phase of wind farm</b> (ref. Pre-/planning)           |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| Construction   | -0.07*               | 0.04          | 0.01                     | 0.06                 | 0.07                   | 0.06                 | -0.03                | 0.03           | 0.02                     | 0.06                 | 0.01                       | 0.05            |
| Deration   | -0.07**              | 0.03          | -0.15***                 | 0.04                 | 0.21***                | 0.04                 | -0.02                | 0.02           | -0.14***                 | 0.04                 | 0.16***                    | 0.04            |
| wareness of renewable energy initiatives (yes, %)        |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| now of a sustainable energy community in the area        | -0.05*               | 0.03          | -0.03                    | 0.04                 | 0.08*                  | 0.04                 | -0.03                | 0.02           | -0.01                    | 0.04                 | 0.04                       | 0.04            |
| vailed of a home energy grant                            | 0.00                 | 0.02          | -0.05                    | 0.04                 | 0.05                   | 0.04                 | 0.01                 | 0.02           | -0.05                    | 0.04                 | 0.04                       | 0.03            |
| nvestments/shares in wind energy                         | -0.04                | 0.04          | 0.05                     | 0.05                 | -0.01                  | 0.05                 | -0.04                | 0.04           | 0.06                     | 0.05                 | -0.02                      | 0.05            |
| Villing to get involved in a community wind farm         | -0.07***             | 0.03          | 0.06*                    | 0.04                 | 0.01                   | 0.04                 | -0.06**              | 0.02           | 0.06*                    | 0.03                 | -0.01                      | 0.03            |
| Vould like to purchase exclusively 'green' electricity   | -0.07***             | 0.02          | -0.04                    | 0.03                 | 0.11***                | 0.03                 | -0.02                | 0.02           | -0.02                    | 0.03                 | 0.05                       | 0.03            |
| trong community spirit in the community                  | -0.02                | 0.03          | -0.06                    | 0.04                 | 0.08**                 | 0.04                 | -0.02                | 0.02           | -0.05                    | 0.04                 | 0.07*                      | 0.04            |
| ndividual characteristics                                |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| emale (ref. male)  | 0.00                 | 0.02          | 0.02                     | 0.03                 | -0.02                  | 0.03                 | 0.03                 | 0.02           | 0.02                     | 0.03                 | -0.05*                     | 0.03            |
| Iomeowner <sup>a</sup>                                   | 0.03                 | 0.03          | -0.03                    | 0.04                 | 0.00                   | 0.04                 | 0.02                 | 0.02           | -0.05                    | 0.04                 | 0.03                       | 0.03            |
| andowner   | 0.02                 | 0.02          | 0.00                     | 0.03                 | -0.02                  | 0.03                 | 0.02                 | 0.02           | -0.01                    | 0.03                 | 0.01                       | 0.03            |
| se (ref 18-34) rowhead                                   | 0.02                 | 0.02          | 0.00                     | 0.00                 | 0.02                   | 0.00                 | 0.00                 | 0.02           | 0.01                     | 0.00                 | 0.01                       | 0.00            |
| 5-54   | 0.06***              | 0.02          | -0.07**                  | 0.03                 | 0.01                   | 0.04                 | 0.01                 | 0.02           | -0.06*                   | 0.03                 | 0.05                       | 0.03            |
| 5-54   | 0.14***              | 0.02          | -0.09*                   | 0.05                 | -0.05                  | 0.05                 | 0.07**               | 0.02           | -0.08*                   | 0.05                 | 0.03                       | 0.05            |
| ducation (ref Drimary intermediate junior certificate)   | 0.14                 | 0.04          | -0.05                    | 0.05                 | -0.05                  | 0.05                 | 0.07                 | 0.05           | -0.00                    | 0.05                 | 0.01                       | 0.05            |
| acondary leaving certificate                             | 0.00                 | 0.03          | 0.02                     | 0.05                 | 0.02                   | 0.05                 | 0.03                 | 0.03           | 0.03                     | 0.05                 | 0.06                       | 0.05            |
| bird level and higher                                    | 0.00                 | 0.03          | 0.02                     | 0.05                 | -0.02                  | 0.05                 | 0.05**               | 0.03           | 0.03                     | 0.05                 | -0.00                      | 0.05            |
| maleyment status (ref. Working full time)                | 0.03                 | 0.03          | 0.03                     | 0.05                 | -0.00                  | 0.03                 | 0.05                 | 0.02           | 0.04                     | 0.05                 | -0.09                      | 0.05            |
| Working port time  | 0.05**               | 0.02          | 0.02                     | 0.04                 | 0.02                   | 0.04                 | 0.02*                | 0.02           | 0.04                     | 0.04                 | 0.00                       | 0.04            |
| totirod  | -0.03                | 0.02          | 0.03                     | 0.04                 | 0.02                   | 0.04                 | -0.03                | 0.02           | 0.04                     | 0.04                 | 0.00                       | 0.04            |
| lot currently working <sup>b</sup>                       | -0.03                | 0.03          | -0.00                    | 0.00                 | 0.09                   | 0.00                 | -0.02                | 0.03           | -0.07                    | 0.00                 | 0.09                       | 0.00            |
| we we have hald in some (ref. (22.240 or under)          | 0.04                 | 0.03          | 0.02                     | 0.04                 | -0.07                  | 0.04                 | 0.01                 | 0.02           | 0.02                     | 0.04                 | -0.03                      | 0.04            |
|  | 0.01                 | 0.02          | 0.00                     | 0.04                 | 0.01                   | 0.04                 | 0.01                 | 0.00           | 0.01                     | 0.04                 | 0.00                       | 0.04            |
| 23,400 - 62,348  | 0.01                 | 0.03          | 0.00                     | 0.04                 | -0.01                  | 0.04                 | -0.01                | 0.02           | 0.01                     | 0.04                 | 0.00                       | 0.04            |
| 62,400 or over   | -0.03                | 0.03          | -0.01                    | 0.04                 | 0.03                   | 0.05                 | -0.03                | 0.03           | -0.01                    | 0.04                 | 0.04                       | 0.04            |
| umber of people in nousenoid (ref. 1 person)             | 0.04                 | 0.00          | 0.00+                    | 0.05                 | 0.00                   | 0.05                 | 0.00                 | 0.00           | 0.00+                    | 0.05                 | 0.07                       | 0.04            |
| people   | -0.06*               | 0.03          | 0.08*                    | 0.05                 | -0.02                  | 0.05                 | -0.02                | 0.02           | 0.08*                    | 0.05                 | -0.06                      | 0.04            |
| i+ people  | -0.06*               | 0.03          | 0.02                     | 0.05                 | 0.04                   | 0.05                 | -0.01                | 0.03           | 0.02                     | 0.05                 | 0.00                       | 0.05            |
| ime living in current community (ref. under 1 year)      | 0.00                 | 0.05          | 0.00                     | 0.00                 | 0.10                   | 0.00                 | 0.00                 | 0.05           | 0.07                     | 0.07                 | 0.10                       | 0.07            |
| -10 years  | -0.03                | 0.05          | -0.09                    | 0.08                 | 0.12                   | 0.08                 | -0.03                | 0.05           | -0.07                    | 0.07                 | 0.10                       | 0.07            |
| 1–20 years   | -0.01                | 0.06          | -0.04                    | 0.08                 | 0.06                   | 0.08                 | -0.03                | 0.05           | -0.03                    | 0.08                 | 0.07                       | 0.07            |
| Over 20 years  | 0.00                 | 0.06          | -0.09                    | 0.08                 | 0.09                   | 0.08                 | -0.02                | 0.05           | -0.07                    | 0.08                 | 0.09                       | 0.07            |
| <b>Rural/Urban area</b> (ref. Rural area)                |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| mall town (population between 1500 and 5000)             | -0.04                | 0.03          | 0.00                     | 0.04                 | 0.04                   | 0.04                 | -0.03                | 0.02           | 0.00                     | 0.04                 | 0.03                       | 0.04            |
| Aid-sized town (population between 5000 and 50,000)      | -0.09***             | 0.02          | 0.04                     | 0.04                 | 0.05                   | 0.04                 | -0.05**              | 0.02           | 0.03                     | 0.04                 | 0.03                       | 0.04            |
| city (population greater than 50,000)                    | -0.06                | 0.05          | -0.07                    | 0.07                 | 0.13*                  | 0.07                 | -0.06                | 0.04           | -0.04                    | 0.07                 | 0.10                       | 0.07            |
| Opinions concerning wind electricity (agree, %)          |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| clean renewable energy source                            |                      |               |                          |                      |                        |                      | -0.04*               | 0.02           | 0.00                     | 0.05                 | 0.04                       | 0.05            |
| ey to achieving carbon reduction commitments             |                      |               |                          |                      |                        |                      | -0.04**              | 0.02           | 0.01                     | 0.04                 | 0.03                       | 0.04            |
| Inreliable because the wind does not always blow         |                      |               |                          |                      |                        |                      | -0.04*               | 0.02           | 0.00                     | 0.04                 | 0.04                       | 0.04            |
| oes more environmental harm than good                    |                      |               |                          |                      |                        |                      | 0.03                 | 0.02           | -0.06                    | 0.04                 | 0.03                       | 0.04            |
| Vind farm developments can negatively impact tourism     |                      |               |                          |                      |                        |                      | 0.03*                | 0.02           | 0.08**                   | 0.03                 | $-0.11^{***}$              | 0.03            |
| here is a role for both onshore and offshore wind energy |                      |               |                          |                      |                        |                      | -0.07***             | 0.02           | -0.04                    | 0.04                 | 0.11***                    | 0.04            |
| Vind farm developments should be moved offshore          |                      |               |                          |                      |                        |                      | 0.01                 | 0.02           | 0.03                     | 0.03                 | -0.04                      | 0.03            |
| this faith developments should be moved subhore          |                      |               |                          |                      |                        |                      |                      |                |                          |                      |                            |                 |
| Vind energy has the potential to create jobs             |                      |               |                          |                      |                        |                      | -0.03*               | 0.02           | 0.02                     | 0.04                 | 0.02                       | 0.04            |

| continued |
|-----------|
| 8         |
| ole       |
| Tal       |

|  | ), :I ləpoM         | Constrained'               |                        |                       |          |                          | I, :Z ləpoW          | 'ull'          |                        |                      |                          |                   |
|--|---------------------|----------------------------|------------------------|-----------------------|----------|--------------------------|----------------------|----------------|------------------------|----------------------|--------------------------|-------------------|
|  | Unwilling<br>(13 %) | ness to accept             | Moderate<br>accept (35 | willingness to<br>(%) | Strong a | willingness to<br>[52 %] | Unwillingı<br>(13 %) | less to accept | Moderate<br>accept (35 | willingness to<br>%) | Strong wil<br>accept (52 | lingness to<br>%) |
|  | AME <sup>a</sup>    | $\mathrm{SE}^{\mathrm{b}}$ | AME                    | SE                    | AME      | SE                       | AME                  | SE             | AME                    | SE                   | AME                      | SE                |
| Very) likely to trust sources of information (%)       |                     |                            |                        |                       |          |                          |                      |                |                        |                      |                          |                   |
| Vebsite run by the developer                           |                     |                            |                        |                       |          |                          | $-0.04^{**}$         | 0.02           | -0.03                  | 0.04                 | 0.07**                   | 0.03              |
| Vebsite run by an anti-wind farm segment               |                     |                            |                        |                       |          |                          | 0.02                 | 0.02           | -0.01                  | 0.03                 | -0.01                    | 0.03              |
| ocal paper   |                     |                            |                        |                       |          |                          | $0.04^{*}$           | 0.02           | -0.01                  | 0.04                 | -0.04                    | 0.03              |
| ocal radio   |                     |                            |                        |                       |          |                          | 0.01                 | 0.02           | -0.06*                 | 0.04                 | 0.05                     | 0.03              |
| lected representative such as councillor/TD            |                     |                            |                        |                       |          |                          | -0.01                | 0.02           | -0.04                  | 0.03                 | 0.05                     | 0.03              |
| ocial media such as Twitter and Facebook               |                     |                            |                        |                       |          |                          | -0.03                | 0.02           | -0.02                  | 0.03                 | 0.04                     | 0.03              |
| community liaison officer from the wind farm developer |                     |                            |                        |                       |          |                          | $-0.05^{**}$         | 0.02           | -0.06                  | 0.04                 | $0.10^{***}$             | 0.03              |
| nformation pack provided by the developer              |                     |                            |                        |                       |          |                          | -0.02                | 0.02           | 0.02                   | 0.04                 | 0.01                     | 0.04              |
| , neighbour who probably knows more about these things |                     |                            |                        |                       |          |                          | 0.01                 | 0.02           | 0.01                   | 0.03                 | -0.02                    | 0.03              |
| Vald Chi <sup>2</sup> (df)                             | 211.63 (6           | (0                         |                        |                       |          |                          | 332.1                | 2 (96)         |                        |                      |                          |                   |
|  | 1109                |                            |                        |                       |          |                          | 1109                 |                |                        |                      |                          |                   |

Standard error of the mean (SE): a measure of sample heterogeneity equal to standard deviation over the square-root of sample size.

Renewable and Sustainable Energy Reviews 189 (2024) 113839

"wind caution" towards commercial development at a local scale, and a preference to address national energy issues differently [109]. For instance, residents unwilling to accept are less likely to agree there is a role for onshore and offshore wind energy. Despite this, residents unwilling to accept generally perceive wind farms as being reliable. Respondents who have doubts regarding the fairness of cost-benefit trade-offs of wind development, particularly with respect to local interests and industries, are generally less willing to accept wind energy development in the community [7,69].

Moreover, residents' degree of trust in sources of information about a new wind farm has a bearing on their acceptance for further development in the community. Residents have different propensities to trust information circulated through the local media. This may be because controversial projects tend to attract greater focus from local stakeholders [50], social media [70] and the mainstream press [22]. For example, respondents unwilling to accept additional wind farms in the community are more likely to trust information circulated within local newspaper coverage whereas respondents with moderate preferences perceive local radio as less trustworthy on average. Notably, Model 2 shows that residents who trust information provided by a developer's website or community liaison officer are more likely to have a strong willingness to accept. Conversely, on average residents who are unwilling to accept additional development in the area are significantly less trusting of these sources of information. This comparison underlines that acceptance is affected by the quality and openness of relationships between different stakeholders involved in the development and operation of a project. Research indicates it is not conducive to relationship-building if wind farm neighbours only learn about projects after they are complete [28], which reinforces the importance and challenges for mediators such as community liaison officers to establish their legitimacy in the consultation process [83].

Finally, Models 1 and 2 show that several controlling sociodemographic variables correlate with acceptance for wind farms proposed within 10 km. It is not well understood precisely how sociodemographic context shapes the acceptance of onshore wind energy, and there is conflicting evidence as to the significance of variables such as gender, income and level of educational attainment [54]. These results show that male respondents living in areas which have a higher population are more likely to be supportive of further wind development in the community. Comparatively, residents who are unwilling to accept further wind development in their community are more likely to be female, to live in rural areas, and to have a third-level education. Residents with a 'moderate' stance towards further wind development are more likely to belong to a two-person household, to be working part-time and to be younger than 35 years old. It is worth noting that respondents who are unwilling to accept further implementation in the area also tend to be older on average, representing double as many 55+ years-olds than the full sample (42 %, Table 6). These findings are consistent with technology acceptance theory [110] and empirical research [40,111] which suggest that age is an important factor in how people perceive the usefulness of innovation and that experience and knowledge have direct and indirect effects on the acceptance of sustainable energy technologies [21]. It has similarly been said that younger generations have a different perception of environmental change and land use arising from 'shifting baseline syndrome' [24] and are more likely to view increased renewable energy development as normal and acceptable [40,112].

In summary, together these results reveal that the majority of residents living within 10 km of an existing wind farm are generally supportive of further onshore wind development in the community but emphasise the ongoing participation and benefit of local stakeholders throughout planning, construction and operation. Hesitancy towards wind farm development can be viewed as part of a natural reaction to change, such that the process of acceptance takes longer for some than for others [17,113]. Nevertheless, the findings support that acceptance generally increases with experience of operating wind farms. Factors

including trust in information, participation with renewable energy and personal values differentially shape this process. Previous studies have revealed a close relationship between participatory justice and community acceptance for wind energy developments [25,27,54]. As the leader of the stakeholder engagement process, the wind farm developer is the focal point for a series of relationships which can collectively reinforce or undermine the legitimacy of a commercial wind farm [83].

Discourses concerning wind energy development cover a variety of motivations for and against further development, for instance a desire to anticipate unforeseen local impacts, as well capturing scepticism or enthusiasm about the potential of wind energy, or the incentives and motivations of actors such as developers [18,114]. Early research named community resistance to the visual impact of wind farms on the natural landscape as a primary impediment to the diffusion of wind energy, second only to economic feasibility constraints [115]. However, a more nuanced picture is emerging. It suggests residents' acceptance for the continued expansion of wind energy is interrogated according to the perceived benefits and impacts of wind energy developments, which have different scales (visible 'local' impacts to provide abstract 'global' benefits) and change over the long-term (since the benefits of both local wind farms and the global energy transition accrue by degrees and over decades) [22]. Within this context, the results suggest that favourable attitudes towards the development of further wind farms in the local community are reinforced through prior experiences which corroborate with residents' expectations for equity and justice, for instance with the realisation of benefits for the community and environment [58]. The findings also suggest that citizen participation in wind farms is a persistent and significant issue affecting the perceived legitimacy of further development [14,25,9]. Procedural issues such as the information about the wind farm, the extent of residents' involvement in the project and the relationships between stakeholders shape the acceptance of wind farms over their lifetime. Though many residents express support for additional wind farms, a comparatively smaller, but still important, proportion living near existing wind farms are not willing to accept further development in the community. This contributes to the research which recognises local concerns about the suitability and fit of new projects within the community as lasting issues for future implementation [41,57,108,46]. The findings provide lessons to support fairness which is particularly relevant at the earliest phases of wind farm development when there is greater likelihood for uncertainty and resistance to change [9,85]. The following section concludes with specific recommendations for policy, practice and future research.

# 5. Conclusion

Local acceptance of renewable energy developments by residents living nearby is a focal point for the decarbonisation of the electricity sector as reiterated by policy [106,116,117], the wind industry [118, 119] and the international press [120,121]. An enhanced understanding of how activities associated with different phases of nearby wind farm development affect residents' acceptance is a crucial consideration for further wind energy deployment [5,48]. This is especially important considering that global renewable capacity targets have increased by 85 % over the last five years and are projected to continue to accelerate within the context of the European energy crisis [4]. Alongside new developments, more than a quarter of the earliest wind farms are approaching the end of their operational phase and many are anticipated to pursue planning applications to extend their operational permits [50,122].

This paper investigates the relationship between wind farm neighbours' level of experience with wind farms and their acceptance for further development in the community. The analysis gives specific focus to wind farm neighbours who have experience of living within 10 km of a wind farm at the planning, construction or operation phase of development. Using a choice experiment which provided 1109 Irish residents with specific proposals for additional wind farm developments in the community, it identifies three latent classes of residents exhibiting distinctly different levels of acceptance. It investigates to what extent willingness to accept is determined by temporal and proximal experiences of wind farms, accounting for awareness and involvement in renewable energy, as well as persistent personal factors such as general sentiments towards onshore windgenerated electricity and trust in information sources concerning wind farm proposals. There is a strong relationship between willingness to accept and awareness of local renewable energy initiatives, trust in information sources, and the perceived fairness of local benefits. These findings guide recommendations for policies concerning wind farm development processes.

#### 5.1. Key findings: Evolution of perceptions and responses

The key findings provide evidence in support of the hypothesis that prior experience with wind farms is linked to greater acceptance for further development in the area [14,15,54]. Residents who live near an operating wind farm [9], or who live beyond the 2 km 'near-neighbour zone' of likely impact [15,84] are on average more supportive of additional wind farm developments in the community.

Of residents living within 10 km of a wind farm, more than half show strong willingness to accept additional wind farm developments in the community. Residents who are willing to accept perceive community engagement across planning, construction and operation to have a high relative importance. Building on earlier research, the paper finds that acceptance for a nearby project relates to residents' appraisal of the collective socioeconomic and environmental benefits of wind electricity in general, as well as more specific preferences for the development of a proposed wind farm in the area [21,29,109]. For example, 13 % of residents are unwilling to accept a proposed wind farm, give greater emphasis to the perceived costs of development, and are generally less convinced of potential benefits [21]. These residents are older on average and are more sensitive to proximity and visual attributes of specific wind farm development proposals as well as socially or environmentally contentious aspects wind energy, such as the perceived negative impact on tourism [18].

Additionally, two key factors differentiate strong support from weaker attitudes towards wind farms proposed in a community. Firstly, prior awareness or involvement in local energy initiatives such as sustainable energy communities, a sense of community spirit and a desire use 'green' electricity contribute to willingness to accept. Secondly, trust in sources of information about a project, particularly a developer's website or community liaison officer, is strongly related to acceptance. This contributes towards a growing body of work that grassroots energy initiatives and local relationships can spark a supportive collective interest in renewable energy and are less likely to engage with information provided about a project or developer [14,28,123]. Together these findings have implications for policy supports and the topic warrants further investigation.

# 5.2. Recommendations for policy and future research

It is in the interests of effective wind projects that developers and policymakers recognise that uncertainty, partiality and resistance are reasonable responses towards wind farm development proposals [52]. Even as wind energy establishes itself as a viable source of wind electricity, social buy-in to new projects is not guaranteed, and it is to be expected that acceptance within the nearest vicinity of a wind farm is weaker at the earliest and most uncertain phases of project development [9,23]. Surveyed wind farm neighbours place high importance on engagement and local participation throughout planning, construction and operation. The longer-term acceptance of a project depends on trust in information provided by a developer at time when a new wind farm is proposed for development. This supports frameworks which recommend that developers should aim to engage, understand and address local residents' concerns as a starting point for fair development procedures [7].

Since the earliest wind farm proposals, there have been numerous calls for sensitivity to local interests during siting and for deeper levels of citizen influence on the decision-making process [57,107]. Nevertheless, evidence suggests that the impacts of wind farms, whether expected or experienced, can be an important aspect of residents' acceptance [11]. Perceived impacts are affective issues and can exacerbate uncertainty during planning [124] and extend well beyond a wind farm's perceptible impact [15,84]. Critically for developers, the first point contact with local residents can determine longer-lasting attitudes towards the wind farm [28]. The energy transition increasingly requires efforts to build working relationships between renewable energy developers and local stakeholders, often over a longer time frame and in a more deliberative fashion than once-off compensation and communication campaigns [125]. For example, this study finds community liaison officers can provide an important intermediary role, however it is not guaranteed that they will be trusted by members of the community. The results also suggest that it is valuable for developers to communicate a proposed development across multiple channels, including local radio, and to undertake in-person consultation. This can help to gather feedback at the early stages of planning and to encourage the representation of diverse stakeholder interests and credible information-sharing during decision-making.

Supportive attitudes towards further wind farm development are linked to strong preferences for community-owned wind farms and citizen investment opportunities. Therefore, it is in the interests of a just energy transition to develop enabling frameworks for community participation in renewable energy initiatives. This can support social capital through familiarisation [14], experience- and knowledge-building [123,89] and help foster psychological ownership of a local energy transition [98].

Considering the cross-sectional nature of this research, there is interesting scope to build upon these findings and deepen investigation into the longitudinal nature of wind farm acceptance over time. Future research efforts could conduct repeated focus groups supplemented by a repeat survey at later points in time. Transition management strategies could be tailored to engage local communities during siting, construction and operation with deeper investigation into the effects of 'learning-by-doing' [126] and local culture [108]. A valuable area of ongoing

research aims to contextualise stakeholder concerns, for instance the site-specific effects of wind farm implementation on local tourism [74, 127,128]. Similarly, an important area of research takes into consideration the proximity-dependent impacts experienced by residents (such as noise emissions [15] or visibility [24]) and aims to investigate to what extent these affect respondents' acceptance of wind farms. Studies of specific wind farm locations including topographical context could also provide further insight into the relationship between proximity, timing of development, and acceptance. For example, there is evidence to suggest with appropriate regulation to minimise impacts, the planning process has primary importance for wind energy acceptance [11]. There is also a need to focus on jurisdictions with pressing concerns for electricity supply, security or affordability to understand the differing local requirements for, and perceptions of, renewable energy infrastructure [129].

# CRediT authorship contribution statement

Julia le Maitre: Conceptualization, Methodology, Software, Validation, Formal analysis, Data curation, Investigation, Writing – original draft, Writing – review & editing, Visualization, Funding acquisition. Geraldine Ryan: Conceptualization, Supervision, Funding acquisition, Investigation, Data curation. Bernadette Power: Conceptualization, Supervision, Funding acquisition, Investigation, Data curation.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Data availability

The data that has been used is confidential.

#### Acknowledgements

This research was supported by funding from Horizon Europe Marie Skłodowska-Curie Actions consortium MISTRAL, 'Multi-sectoral approaches to Innovative Skills Training for Renewable energy And sociaL acceptance' [813837]; the Irish Research Council and Environmental Protection Agency [GOIPG/2022/1722]; and Sustainable Energy Authority Ireland project Co-Wind, 'Community Engagement in Wind Energy: Innovative approaches to achieving a social license' [18/RDD/ 281]. The authors gratefully acknowledge these grants and responsibility for this publication. Thank you to the focus group participants and expert advisors whose experience helped to guide the development of this research to its completion. Our thanks also to the Sustainable Energy Authority Ireland Wind Research and Technology team for their technical support and guidance, and to all researchers involved in the St Gallen REMforum 2022 and the 'Dynamics of Social Acceptance' Special Issue workshop for the valuable discussions and insight.

# Appendix

| 1 Latent class           | 1       | 2      | 3    | 4     | 5     | 6     | 7     | 8     | 9     | 10    | 11    | 12    | 13    | 14    | 15     | 16     | 17     | 18     | 19     | 20     | 21     | 22      | 23     | 24     | 25     | 26    | 27    | 28     | 29    | 30     | 31    | 32 :    | 33    | 34     | 35   | 36 37 |
|--------------------------|---------|--------|------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|--------|--------|--------|--------|--------|--------|--------|---------|--------|--------|--------|-------|-------|--------|-------|--------|-------|---------|-------|--------|------|-------|
| 2 Gender                 | 1.00    | _      | -    |       | -     |       |       |       |       |       |       |       |       |       |        |        |        |        |        |        | _      |         |        |        |        |       |       |        |       |        | -     |         |       | -      |      |       |
| 3 Age                    | -0.05   | 1.00   |      |       |       |       |       |       |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 4 Education              | 0.10    | 0.02   | 1.00 |       |       |       |       |       |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 5 Employment             | -0.11   | 0.51 - | 0.06 | 1.00  |       |       |       |       |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 6 Homeowner              | -0.07   | 0.22   | 0.00 | 0.17  | 1.00  |       |       |       |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 7 Landowner              | -0.09   | 0.14 - | 0.08 | -0.08 | 0.25  | 1.00  |       |       |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 8 Household size         | 0.02 .  | 0.25 - | 0.06 | -0.23 | 0.03  | 0.07  | 1.00  |       |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 9 Income                 | -0.09   | 0.07   | 0.17 | -0.12 | 0.22  | 0.12  | 0.20  | 1.00  |       |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 10 Years in community    | 0.06    | 0.21   | 0.02 | 0.17  | 0.22  | -0.04 | 0.06  | 0.01  | 1.00  |       |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 11 Rural/Urban           | -0.06   | 0.11   | 0.01 | -0.02 | -0.16 | -0.16 | 0.02  | 0.01  | -0.09 | 1.00  |       |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 12 NUTS3                 | 0.01    | 0.08   | 0.07 | 0.01  | -0.02 | -0.08 | -0.03 | -0.02 | -0.04 | -0.03 | 1.00  |       |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 13 Proximity             | 0.05    | 0.20   | 0.08 | 0.13  | -0.09 | -0.26 | 0.00  | -0.04 | 0.08  | 0.22  | 0.04  | 1.00  |       |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 14 Project phase         | 0.00    | 0.27   | 0.02 | 0.15  | -0.06 | -0.25 | 0.03  | -0.10 | 0.03  | 0.19  | 0.04  | 0.77  | 1.00  |       |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 15 Attitude: Unreliable  | -0.16   | 0.09 - | 0.09 | -0.02 | 0.09  | 0.21  | 0.05  | -0.02 | -0.07 | 0.01  | -0.05 | -0.22 | -0.19 | 1.00  |        |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 16 Attitude: Clean       | 0.08    | 0.01   | 0.07 | 0.04  | -0.06 | -0.12 | 0.00  | 0.00  | -0.01 | 0.05  | 0.07  | 0.16  | 0.08  | -0.26 | 1.00   |        |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 17 Attitude: Harm        | -0.13 - | 0.12 - | 0.10 | -0.07 | 0.05  | 0.18  | 0.04  | -0.03 | -0.04 | -0.01 | -0.11 | -0.22 | -0.17 | 0.43  | -0.32  | 1.00   |        |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 18 Attitude: Commitments | -0.04 - | 0.01   | 0.06 | 0.02  | -0.02 | -0.05 | -0.01 | 0.03  | -0.03 | 0.06  | 0.04  | 0.06  | 0.03  | -0.15 | 0.39 - | 0.17   | 1.00   |        |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 19 Attitude: Tourism     | -0.10 - | 0.03 - | 0.01 | 0.00  | 0.11  | 0.17  | 0.01  | 0.00  | -0.01 | -0.05 | -0.02 | -0.15 | -0.13 | 0.35  | -0.18  | 0.37 - | 0.23   | 1.00   |        |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 20 Attitude: Jobs        | 0.04 -  | 0.01   | 0.09 | 0.00  | -0.03 | -0.06 | -0.01 | 0.05  | 0.00  | 0.04  | 0.02  | 0.08  | 0.03  | -0.13 | 0.34 - | 0.16   | 0.32 - | 0.18   | 1.00   |        |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 21 Attitude: Discord     | -0.06   | 0.13   | 0.01 | 0.06  | 0.18  | 0.10  | -0.07 | 0.03  | 0.07  | -0.10 | -0.03 | -0.10 | -0.10 | 0.22  | -0.10  | 0.22 - | -0.08  | 0.36 - | 0.10   | 1.00   |        |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 22 Attitude: Offshore    | -0.12   | 0.05 - | 0.06 | 0.08  | 0.14  | 0.14  | -0.01 | 0.01  | 0.01  | -0.10 | -0.02 | -0.09 | -0.07 | 0.24  | -0.11  | 0.25 - | 0.08   | 0.33 - | 0.07   | 0.23   | 1.00   |         |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 23 Attitude: On/offshore | 0.01    | 0.00   | 0.09 | 0.05  | -0.03 | -0.09 | -0.02 | 0.07  | 0.00  | 0.01  | 0.03  | 0.04  | 0.00  | -0.10 | 0.31 . | 0.14   | 0.34 - | 0.14   | 0.29 - | 0.06 - | 0.13   | 1.00    |        |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 24 Green electricity     | -0.03   | 0.05   | 0.04 | 0.04  | 0.04  | 0.04  | 0.02  | 0.04  | -0.02 | -0.01 | 0.00  | -0.07 | -0.07 | -0.09 | 0.13 · | 0.10   | 0.19 - | 0.11   | 0.08 - | 0.03 - | 0.01   | 0.12    | 1.00   |        |        |       |       |        |       |        |       |         |       |        |      |       |
| 25 Availed of grant      | -0.10   | 0.06 - | 0.09 | 0.01  | 0.17  | 0.18  | 0.05  | 0.05  | -0.03 | 0.04  | -0.08 | -0.14 | -0.10 | 0.15  | -0.03  | 0.13   | 0.05   | 0.06 - | 0.01   | 80.0   | 0.06 - | -0.02   | 0.12 1 | .00    |        |       |       |        |       |        |       |         |       |        |      |       |
| 26 Investments / shares  | -0.17   | 0.27 - | 0.14 | -0.10 | 0.06  | 0.29  | 0.12  | 0.04  | -0.14 | 0.09  | -0.07 | -0.24 | -0.17 | 0.23  | -0.06  | 0.26   | 0.05   | 0.12 - | 0.01   | 0.03   | 0.08 - | 0.03    | 0.07 0 | .32 1  | 1.00   |       |       |        |       |        |       |         |       |        |      |       |
| 27 Local WF involvement  | -0.13 - | 0.02 - | 0.03 | 0.01  | 0.00  | 0.07  | 0.11  | 0.00  | -0.11 | 0.14  | -0.01 | 0.08  | 0.17  | 0.04  | 0.04   | 0.00   | 0.08 - | 0.03   | 0.03 - | 0.03   | 0.00   | 0.05    | 0.24 0 | .21 0  | 0.23 1 | .00   |       |        |       |        |       |         |       |        |      |       |
| 28 SEC in area           | -0.14 · | 0.29 - | 0.10 | -0.15 | 0.02  | 0.24  | 0.08  | 0.06  | -0.17 | 0.09  | -0.07 | -0.28 | -0.20 | 0.20  | -0.09  | 0.18   | 0.05   | 0.08 - | 0.03   | 0.02   | 0.05 - | -0.02   | 0.07 0 | .31 (  | 0.47 0 | .22 1 | .00   |        |       |        |       |         |       |        |      |       |
| 29 Info: online (dev.)   | 0.06 -  | 0.20 - | 0.02 | -0.16 | -0.10 | 0.06  | 0.08  | 0.00  | -0.11 | 0.05  | 0.03  | -0.04 | -0.05 | 0.02  | 0.11   | 0.02   | 0.17 - | 0.12   | 0.12 - | 0.17 - | 0.11   | 0.15    | 0.11 0 | .05 0  | 0.13 0 | .08 0 | .16 1 | .00    |       |        |       |         |       |        |      |       |
| 30 Info: online (anti.)  | -0.06 - | 0.09 - | 0.09 | -0.09 | 0.00  | 0.10  | 0.06  | -0.06 | -0.11 | 0.03  | 0.02  | -0.14 | -0.08 | 0.23  | -0.11  | 0.24 - | 0.02   | 0.15 - | 0.11   | 80.0   | 0.13 - | -0.10 - | 0.02 0 | .13 (  | 0.24 0 | .08 0 | .26 0 | .14 1. | .00   |        |       |         |       |        |      |       |
| 31 Info: newspaper       | -0.02   | 0.01   | 0.00 | 0.01  | -0.01 | -0.06 | 0.01  | -0.02 | 0.00  | 0.08  | 0.03  | 0.07  | 0.06  | -0.01 | 0.12 . | 0.04   | 0.12 - | 0.03   | 0.11 - | 0.04 - | 0.02   | 0.15    | 0.09 0 | .02 -0 | 0.01 0 | .09 0 | .02 0 | .26 0  | .12 1 | .00    |       |         |       |        |      |       |
| 32 Info: radio           | -0.01   | 0.01   | 0.01 | 0.00  | -0.02 | -0.07 | -0.02 | -0.01 | -0.03 | 0.06  | 0.02  | 0.07  | 0.07  | -0.02 | 0.14 . | 0.06   | 0.12 - | 0.07   | 0.14 - | 0.07 - | 0.01   | 0.12    | 0.09 0 | .02 (  | 0.01 0 | .08 0 | .00 0 | .23 0. | .11 0 | .62 1. | .00   |         |       |        |      |       |
| 33 Info: elected rep     | -0.02 - | 0.15 - | 0.01 | -0.07 | -0.04 | 0.03  | 0.12  | 0.02  | -0.07 | 0.06  | 0.00  | -0.04 | -0.02 | 0.04  | 0.07   | 0.03   | 0.12 - | 0.05   | 0.06 - | 0.08 - | 0.04   | 0.08    | 0.08 0 | .07 0  | 0.13 0 | .10 0 | .14 0 | .34 0. | .24 0 | .29 0. | 29 1  | .00     |       |        |      |       |
| 34 Info: Social media    | -0.04 · | 0.26 - | 0.07 | -0.16 | -0.10 | 0.08  | 0.11  | -0.04 | -0.16 | 80.0  | -0.03 | -0.12 | -0.11 | 0.17  | -0.05  | 0.18   | 0.05   | 0.04   | 0.01 - | 0.05   | 0.05   | 0.01 -  | 0.02 0 | .10 0  | 0.25 0 | .05 0 | .25 0 | .29 0. | .35 0 | .19 0. | .18 0 | .31 1.0 | 00    |        |      |       |
| 35 Info: CLO             | 0.07 -  | 0.10 - | 0.02 | -0.06 | -0.07 | -0.03 | 0.08  | -0.01 | -0.03 | 0.08  | 0.00  | 0.07  | 0.05  | -0.09 | 0.14   | 0.06   | 0.20 - | 0.20   | 0.15 - | 0.15 - | 0.13   | 0.17    | 0.12 0 | .04 0  | 0.04 0 | .10 0 | .08 0 | .46 0  | .07 0 | .29 0. | 27 0  | .30 0.2 | 20 1. | .00    |      |       |
| 36 Info: leaflet         | 0.03 ·  | 0.12 - | 0.01 | -0.05 | -0.07 | -0.02 | 0.06  | -0.02 | -0.05 | 0.06  | 0.01  | 0.05  | 0.04  | -0.05 | 0.18 · | 0.11   | 0.21 - | 0.18   | 0.15 - | 0.18 - | 0.13   | 0.16    | 0.15 0 | .02 0  | 0.07 0 | .12 0 | .10 0 | .54 0  | .07 0 | .24 0. | 25 0  | .29 0.2 | 22 0. | .59 1. | 00   |       |
| 37 Info: neighbour       | -0 10 - | 0.04 - | 0 10 | -0.02 | -0.04 | 0.03  | 0.03  | -0.02 | -0.05 | 0.06  | -0.02 | -0.04 | 0.00  | 0 15  | -0.01  | 0 10   | 0.08   | 0.03 - | 0.01   | 0.02   | 0.07 - | 0.03    | 0 02 0 | 09 0   | 0.14 0 | 06 0  | 16.0  | 13.0   | 28.0  | 24 0   | 25 0  | 26 0 3  | 35 0  | 16 0   | 16 1 | 00    |

Appendix Fig. A1. Spearman's rank correlation matrix of variables with Bonferroni adjustment

# Appendix Table A1

Part-worth utilities and standard error (SE) of the aggregate model

| Attributes                               | Levels (trade-offs)   | Part-worth utility | SE   |
|--|---|--------------------|------|
| Distance to the wind farm                | Outside immediate community (3–10 km)                                     | 0.66               | 0.03 |
|  | Close to community (2–3 km)   | 0.34               | 0.01 |
|  | Close to community (1–2 km)   | -0.26              | 0.01 |
|  | Near neighbour (within 1 km radius)                                       | -0.74              | 0.03 |
| Visual impact                            | Yes   | -0.16              | 0.01 |
|  | No  | 0.16               | 0.01 |
| Engagement during site selection         | One-on-one with a community liaison officer                               | 0.23               | 0.02 |
|  | Meeting of a small group of people living very close to the proposed site | 0.16               | 0.01 |
|  | Town hall style meeting that includes the wider community                 | 0.23               | 0.02 |
|  | You receive a leaflet from the developer                                  | -0.32              | 0.02 |
|  | The developer uses local radio and newspaper                              | -0.30              | 0.01 |
| Community liaison officer                | Yes   | 0.27               | 0.01 |
|  | No  | -0.27              | 0.01 |
| Governance of the community benefit fund | Local authority   | -0.23              | 0.01 |
|  | Wind farm developer with community input                                  | -0.14              | 0.02 |
|  | Local enterprise development company                                      | 0.04               | 0.02 |
|  | Local development company with community input                            | 0.05               | 0.02 |
|  | A committee composed of community members                                 | 0.29               | 0.01 |
| Sharing of the community benefit fund    | €130,000 to community projects and €50,000 to near-neighbours             | -0.13              | 0.02 |
|  | €100,000 to community projects and €80,000 to near-neighbours             | 0.07               | 0.01 |
|  | €80,000 to community projects and €100,000 to near-neighbours             | 0.05               | 0.01 |
| Opportunity for citizens to invest       | Citizens must live close to the windfarm (within a radius of 10 km)       | 0.40               | 0.01 |
|  | Citizens can live anywhere in Ireland                                     | -0.04              | 0.01 |
|  | There is no opportunity to invest   | -0.36              | 0.01 |
| Developer                                | Local landowners/farmers (30 % added cost)                                | 0.19               | 0.01 |
|  | Local landowners/farmers (50 % added cost)                                | -0.18              | 0.02 |
|  | A community sustainable energy group (30 % added cost}                    | 0.22               | 0.01 |
|  | A community sustainable energy group (50 % added cost)                    | -0.39              | 0.01 |
|  | A new venture for an existing community network group (30 % added cost)   | 0.13               | 0.01 |
|  | A new venture for an existing community network group (50 % added cost)   | -0.20              | 0.02 |
|  | Joint developer/community (20 % added cost)                               | 0.15               | 0.01 |
|  | Irish semi state companies  | 0.21               | 0.02 |
|  | Private companies/developers  | -0.14              | 0.02 |
| Status quo                               | Dual-response   | -1.86              | 0.13 |

Note: Within an attribute, more strongly positive coefficients indicate increasing preference compared to the mean (zero), while more negative coefficients mean that these features of wind development tended to have comparatively less influence on respondents' preferences.

#### Appendix Table A2

Average acceptance of a wind farm for segments at different phases of development and proximities

| Proximity <sup>a</sup> | Phase of development | Average frequency of acceptance (%) | SE         | [95 % CI]        | Proportion of wind farm neighbours within 10 km (n = 1109) $$ |
|------------------------|----------------------|-------------------------------------|------------|------------------|---|
| Within 2 km***         | (Pre-planning)       | 47.5 %                              | 5.9 %      | [35.9 %; 59.2 %] | 4.5 %   |
|                        | Construction         | 63.5 %                              | 8.6 %      | [46.5 %; 80.6 %] | 1.1 %   |
|                        | Operation            | 75.0 %                              | 3.2 %      | [68.6 %; 81.4 %] | 0.5 %   |
| 2-5 km**               | (Pre-planning)       | 68.1 %                              | 4.0 %      | [60.2 %; 75.9 %] | 7.2 %   |
|                        | Construction         | 79.0 %                              | 3.8 %      | [71.5 %; 86.4 %] | 3.9 %   |
|                        | Operation            | 80.8 %                              | 1.7 %      | [77.4 %; 84.2 %] | 28.0 %  |
| 5-10 km                | (Pre-planning)       | 70.8 %                              | 3.0 %      | [64.9 %; 76.7 %] | 10.1 %  |
|                        | Construction         | 81.9 %                              | 3.1 %      | [75.8 %; 87.9 %] | 5.5 %   |
|                        | Operation            | 76.8 %                              | $1.8 \ \%$ | [73.3 %; 80.4 %] | 29.0 %  |

<sup>a</sup> Within-segment significance H(2): \*p < 0.1, \*\*p < 0.05, \*\*\*p < 0.01.

#### References

- Hallan C, González A. Adaptive responses to landscape changes from onshore wind energy development in the Republic of Ireland. Land Use Pol 2020;97: 104751.
- [2] Ladenburg J, Krause G. Local attitudes towards wind power: the effect of prior experience. From turbine to wind farms: Technical requirements and spin-off products 2011:3–14.
- [3] IEA. World. Energy outlook. 2021.
- [4] IEA. Renewables. 2022.
- [5] Rudolph D, Clausen LT. Getting used to it, but...? Rethinking the elusive U-curve of acceptance and post-construction assumptions. In: Anonymous A critical approach to the social acceptance of renewable energy infrastructures. Springer; 2021. p. 63–81.
- [6] Sovacool BK, Kim J, Yang M. The hidden costs of energy and mobility: a global meta-analysis and research synthesis of electricity and transport externalities. Energy Res Social Sci 2021;72:101885.
- [7] Petrova MA. From NIMBY to acceptance: toward a novel framework VESPA for organizing and interpreting community concerns. Renew Energy 2016;86: 1280–94.

- [8] Windemer R. Considering time in land use planning: an assessment of end-of-life decision making for commercially managed onshore wind schemes. Land Use Pol 2019;87:104024.
- [9] Wolsink M. Wind power implementation: the nature of public attitudes: equity and fairness instead of 'backyard motives'. Renew Sustain Energy Rev 2007;11: 1188–207.
- [10] Huijts NM, de Vries G, Molin EJ. A positive shift in the public acceptability of a low-carbon energy project after implementation: the case of a hydrogen fuel station. Sustainability 2019;11:2220.
- [11] Hübner G, Leschinger V, Müller FJY, Pohl J. Broadening the social acceptance of wind energy – an Integrated Acceptance Model. Energy Pol 2023;173:113360.
- [12] Wolsink M. Social acceptance, lost objects, and obsession with the 'public'—the pressing need for enhanced conceptual and methodological rigor. Energy Res Social Sci 2019;48:269–76.
- [13] Ellis G, Schneider N, Wüstenhagen R. Dynamics of social acceptance of renewable energy: an introduction to the concept. Energy Policy 2023;181:113706.
- [14] Bauwens T, Devine-Wright P. Positive energies? An empirical study of community energy participation and attitudes to renewable energy. Energy Pol 2018;118: 612–25.
- [15] Dällenbach N, Wüstenhagen R. How far do noise concerns travel? Exploring how familiarity and justice shape noise expectations and social acceptance of planned wind energy projects. Energy Res Social Sci 2022;87:102300.

#### J. le Maitre et al.

- [16] Jones L, Halstead F, Parsons KJ, Le H, Bui TH, Hackney CR, et al. 2020-Vision: understanding climate (in) action through the emotional lens of loss. J British Acad 2021;40:29–68.
- [17] Alexander KA, Wilding TA, Jacomina Heymans J. Attitudes of Scottish Fishers towards marine renewable energy. Mar Pol 2013;37:239–44.
- [18] Cashmore M, Rudolph D, Larsen SV, Nielsen H. International experiences with opposition to wind energy siting decisions: lessons for environmental and social appraisal. J Environ Plann Manag 2019;62:1109–32.
- [19] Küpers S, Batel S. Time, history and meaning-making in research on people's relations with renewable energy technologies (RETs) – a conceptual proposal. Energy Pol 2023;173:113358.
- [20] Corden A, Millar J. Time and change: a review of the qualitative longitudinal research literature for social policy. Soc Pol Soc 2007;6:583–92.
- [21] Huijts NMA, Molin EJE, Steg L. Psychological factors influencing sustainable energy technology acceptance: a review-based comprehensive framework. Renew Sustain Energy Rev 2012;16:525–31.
- [22] Warren CR, Lumsden C, O'Dowd S, Birnie RV. 'Green on green': public perceptions of wind power in Scotland and Ireland. J Environ Plann Manag 2005; 48:853–75.
- [23] Wilson GA, Dyke SL. Pre- and post-installation community perceptions of wind farm projects: the case of Roskrow Barton (Cornwall, UK). Land Use Pol 2016;52: 287–96.
- [24] Ladenburg J. Visual impact assessment of offshore wind farms and prior experience. Appl Energy 2009;86:380–7.
- [25] Baxter J, Walker C, Ellis G, Devine-Wright P, Adams M, Fullerton RS. Scale, history and justice in community wind energy: an empirical review. Energy Res Social Sci 2020;68:101532.
- [26] Tabi A, Wüstenhagen R. Keep it local and fish-friendly: social acceptance of hydropower projects in Switzerland. Renew Sustain Energy Rev 2017;68:763–73.
- [27] Upham P, Oltra C, À Boso. Towards a cross-paradigmatic framework of the social acceptance of energy systems. Energy Res Social Sci 2015;8:100–12.
- [28] Motosu M, Maruyama Y. Local acceptance by people with unvoiced opinions living close to a wind farm; a case study from Japan. Energy Pol 2016;91:362–70.
- [29] Mills SB, Bessette D, Smith H. Exploring landowners' post-construction changes in perceptions of wind energy in Michigan. Land Use Pol 2019;82:754–62.
- [30] Sharpton T, Lawrence T, Hall M. Drivers and barriers to public acceptance of future energy sources and grid expansion in the United States. Renew Sustain Energy Rev 2020;126:109826.
- [31] Sovacool BK, Griffiths S. The cultural barriers to a low-carbon future: a review of six mobility and energy transitions across 28 countries. Renew Sustain Energy Rev 2020;119:109569.
- [32] Vuichard P. Electrifying the company car: identifying hard and soft barriers among fleet managers in Switzerland. Energy Res Social Sci 2021;77:102098.[33] Brennan N, Van Rensburg TM. Public preferences for wind farms involving
- electricity trade and citizen engagement in Ireland. Energy Pol 2020;147:111872. [34] González A. Daly G. Gleeson J. Congested spaces, contested scales – a review of
- spatial planning for wind energy in Ireland. Landsc Urban Plann 2016;145:12–20.
- [35] Stadelmann-Steffen I, Dermont C. Acceptance through inclusion? Political and economic participation and the acceptance of local renewable energy projects in Switzerland. Energy Res Social Sci 2021;71:101818.
- [36] Salak B, Kienast F, Olschewski R, Spielhofer R, Wissen Hayek U, Grêt-Regamey A, et al. Impact on the perceived landscape quality through renewable energy infrastructure. A discrete choice experiment in the context of the Swiss energy transition. Renew Energy 2022;193:299–308.
- [37] Knauf J. Can't buy me acceptance? Financial benefits for wind energy projects in Germany. Energy Pol 2022;165:112924.
- [38] Vuichard P, Broughel A, Wüstenhagen R, Tabi A, Knauf J. Keep it local and birdfriendly: exploring the social acceptance of wind energy in Switzerland, Estonia, and Ukraine. Energy Res Social Sci 2022;88:102508.
- [39] Lienhoop N. Acceptance of wind energy and the role of financial and procedural participation: an investigation with focus groups and choice experiments. Energy Pol 2018;118:97–105.
- [40] Roddis P, Carver S, Dallimer M, Ziv G. Accounting for taste? Analysing diverging public support for energy sources in Great Britain. Energy Res Social Sci 2019;56: 101226.
- [41] Naumann M, Rudolph D. Conceptualizing rural energy transitions: energizing rural studies, ruralizing energy research. J Rural Stud 2020;73:97–104.
- [42] Clausen LT, Rudolph D. Renewable energy for sustainable rural development: synergies and mismatches. Energy Pol 2020;138:111289.
- [43] Copena D, Simón X. Wind farms and payments to landowners: opportunities for rural development for the case of Galicia. Renew Sustain Energy Rev 2018;95: 38–47.
- [44] Rudolph D, Kirkegaard J, Lyhne I, Clausen N, Kørnøv L. Spoiled darkness? Sense of place and annoyance over obstruction lights from the world's largest wind turbine test centre in Denmark. Energy Res Social Sci 2017;25:80–90.
- [45] Müller FJY, Leschinger V, Hübner G, Pohl J. Understanding subjective and situational factors of wind turbine noise annoyance. Energy Pol 2023;173: 113361.
- [46] Russell A, Firestone J. What's love got to do with it? Understanding local cognitive and affective responses to wind power projects. Energy Res Social Sci 2021;71:101833.
- [47] Cousse J, Wüstenhagen R, Schneider N. Mixed feelings on wind energy: affective imagery and local concern driving social acceptance in Switzerland. Energy Res Social Sci 2020;70:101676.
- [48] Wolsink M. Social acceptance revisited: gaps, questionable trends, and an auspicious perspective. Energy Res Social Sci 2018;46:287–95.

#### Renewable and Sustainable Energy Reviews 189 (2024) 113839

- [49] Cass N, Walker G. Emotion and rationality: the characterisation and evaluation of opposition to renewable energy projects. Emotion, Space and Soc 2009;2:62–9.
- [50] Windemer R. Acceptance should not be assumed. How the dynamics of social acceptance changes over time, impacting onshore wind repowering. Energy Pol 2023;173:113363.
- [51] Eltham DC, Harrison GP, Allen SJ. Change in public attitudes towards a Cornish wind farm: implications for planning. Energy Pol 2008;36:23–33.
- [52] Batel S. A critical discussion of research on the social acceptance of renewable energy generation and associated infrastructures and an agenda for the future. J Environ Policy Plann 2018;20:356–69.
- [53] Ellis G, Ferraro G. The Social Acceptance of Wind Energy: where we stand and the path ahead. European Commission, Joint Research Centre; 2017.
- [54] Rand J, Hoen B. Thirty years of North American wind energy acceptance research: what have we learned? Energy Res Social Sci 2017;29:135–48.
- [55] Hoen B, Firestone J, Rand J, Elliot D, Hübner G, Pohl J, et al. Attitudes of U.S. Wind turbine neighbors: analysis of a nationwide survey. Energy Pol 2019;134: 110981.
- [56] Swofford J, Slattery M. Public attitudes of wind energy in Texas: local communities in close proximity to wind farms and their effect on decisionmaking. Energy Pol 2010;38:2508–19.
- [57] Devine-Wright P, Howes Y. Disruption to place attachment and the protection of restorative environments: a wind energy case study. J Environ Psychol 2010;30: 271–80.
- [58] Baxter J, Morzaria R, Hirsch R. A case-control study of support/opposition to wind turbines: perceptions of health risk, economic benefits, and community conflict. Energy Pol 2013;61:931–43.
- [59] Peri E, Becker N, Tal A. What really undermines public acceptance of wind turbines? A choice experiment analysis in Israel. Land Use Pol 2020;99:105113.
- [60] Meyerhoff J. Do turbines in the vicinity of respondents' residences influence choices among programmes for future wind power generation? J Choice Model 2013;7:58–71.
- [61] van Rijnsoever FJ, van Mossel A, Broecks KPF. Public acceptance of energy technologies: the effects of labeling, time, and heterogeneity in a discrete choice experiment. Renew Sustain Energy Rev 2015;45:817–29.
- [62] le Maitre J, Ryan G, Power B, O'Connor E. Empowering onshore wind energy: a national choice experiment on financial benefits and citizen participation. Energy Pol 2023;173:113362.
- [63] Brennan N, Van Rensburg T M, Morris C. Public acceptance of large-scale wind energy generation for export from Ireland to the UK: evidence from Ireland. J Environ Plann Manag 2017;60:1967–92.
- [64] Van Rensburg TM, Kelley H, Jeserich N. What influences the probability of wind farm planning approval: evidence from Ireland. Ecol Econ 2015;111:12–22.
- [65] EirGrid. RESS 1 final auction results. 2020.
- [66] EirGrid. RESS 2 provisional auction results. 2022.
- [67] Sirr G, Power B, Ryan G, Eakins J, O'Connor E, le Maitre J. An analysis of the factors affecting Irish citizens' willingness to invest in wind energy projects. Energy Pol 2023;173:113364.
- [68] Groth TM, Vogt CA. Rural wind farm development: social, environmental and economic features important to local residents. Renew Energy 2014;63:1–8.
- [69] Slattery MC, Johnson BL, Swofford JA, Pasqualetti MJ. The predominance of economic development in the support for large-scale wind farms in the U.S. Great Plains. Renew Sustain Energy Rev 2012;16:3690–701.
- [70] Borch K, Munk AK, Dahlgaard V. Mapping wind-power controversies on social media: facebook as a powerful mobilizer of local resistance. Energy Pol 2020;138: 111223.
- [71] Firestone J, Hirt C, Bidwell D, Gardner M, Dwyer J. Faring well in offshore wind power siting? Trust, engagement and process fairness in the United States. Energy Research & Social Science 2020;62:101393.
- [72] Hyland M, Bertsch V. The Role of Community Involvement Mechanisms in Reducing Resistance to Energy Infrastructure Development. Ecol Econ 2018;146: 447–74.
- [73] Orme B, Chrzan K. Becoming an expert in conjoint analysis: choice modelling for pros. second ed. ed. Sawtooth Software; 2017.
- [74] Strazzera E, Mura M, Contu D. Combining choice experiments with psychometric scales to assess the social acceptability of wind energy projects: a latent class approach. Energy Pol 2012;48:334–47.
- [75] Tabi A, Hille SL, Wüstenhagen R. What makes people seal the green power deal? — Customer segmentation based on choice experiment in Germany. Ecol Econ 2014;107:206–15.
- [76] Salm S, Hille SL, Wüstenhagen R. What are retail investors' risk-return preferences towards renewable energy projects? A choice experiment in Germany. Energy Pol 2016;97:310–20.
- [77] Boxall PC, Adamowicz WL, Swait J, Williams M, Louviere J. A comparison of stated preference methods for environmental valuation. Ecol Econ 1996;18: 243–53.
- [78] Álvarez-Farizo B, Hanley N. Using conjoint analysis to quantify public preferences over the environmental impacts of wind farms. An example from Spain. Energy Pol 2002;30:107–16.
- [79] Lancaster KJ. A new approach to consumer theory. J Polit Econ 1966;74:132–57.[80] Hanley N, Wright RE, Adamowicz V. Using choice experiments to value the
- environment. Environ Resour Econ 1998;11:413–28. [81] DECC. Community benefit funds - good practice principles handbook. 2021.
- [82] Kerr S, Johnson K, Weir S. Understanding community benefit payments from renewable energy development. Energy Pol 2017;105:202–11.

#### J. le Maitre et al.

#### Renewable and Sustainable Energy Reviews 189 (2024) 113839

- [83] Dwyer J, Bidwell D. Chains of trust: energy justice, public engagement, and the first offshore wind farm in the United States. Energy Res Social Sci 2019;47: 166–76.
- [84] Peri E, Tal A. Is setback distance the best criteria for siting wind turbines under crowded conditions? An empirical analysis. Energy Pol 2021;155:112346.
- [85] Devine-Wright P. Rethinking NIMBYism: the role of place attachment and place identity in explaining place-protective action. J Community Appl Soc Psychol 2009;19:426–41.
- [86] Bidwell D. Thinking through participation in renewable energy decisions. Nature Energy 2016;1:1–4.
- [87] Langer K, Decker T, Menrad K. Public participation in wind energy projects located in Germany: which form of participation is the key to acceptance? Renew Energy 2017;112:63–73.
- [88] Walker C, Baxter J. Procedural justice in Canadian wind energy development: A comparison of community-based and technocratic siting processes. Energy Research & Social Science 2017;29:160–9.
- [89] Bristow G, Cowell R, Munday M. Windfalls for whom? The evolving notion of 'community' in community benefit provisions from wind farms. Geoforum 2012; 43:1108–20.
- [90] Devine-Wright P, Sherry-Brennan F. Where do you draw the line? Legitimacy and fairness in constructing community benefit fund boundaries for energy infrastructure projects. Energy Research & Social Science 2019;54:166–75.
- [91] García J, Cherry T, Kallbekken S. Willingness to accept local wind energy development: does the compensation mechanism matter? Energy Pol 2016;99: 165–73.
- [92] Vuichard P, Stauch A, Dällenbach N. Individual or collective? Community investment, local taxes, and the social acceptance of wind energy in Switzerland. Energy Research & Social Science 2019;58:101275.
- [93] Curtin J, McInerney C, Gallachóir BÓ, Salm S. Energizing local communities—What motivates Irish citizens to invest in distributed renewables? Energy Research & Social Science 2019;48:177–88.
- [94] Jørgensen ML, Anker HT, Lassen J. Distributive fairness and local acceptance of wind turbines: The role of compensation schemes. Energy Policy 2020;138: 111294.
- [95] Goedkoop F, Devine-Wright P. Partnership or placation? The role of trust and justice in the shared ownership of renewable energy projects. Energy Research & Social Science 2016;17:135–46.
- [96] Philpott A, Windemer R. Repower to the people: the scope for repowering to increase the scale of community shareholding in commercial onshore wind assets in Great Britain. Energy Res Social Sci 2022;92:102763.
- [97] Slee B. Is there a case for community-based equity participation in Scottish onshore wind energy production? Gaps in evidence and research needs. Renew Sustain Energy Rev 2015;41:540–9.
- [98] Warren CR, McFadyen M. Does community ownership affect public attitudes to wind energy? A case study from south-west Scotland. Land Use Pol 2010;27: 204–13.
- [99] DCCAE. Renewable electricity support scheme (RESS). 2019.
- [100] McFadden D. The choice theory approach to market research. Market Sci 1986;5: 275–97.
- [101] Salm S. The investor-specific price of renewable energy project risk-A choice experiment with incumbent utilities and institutional investors. Renew Sustain Energy Rev 2018;82:1364–75.
- [102] Ladenburg J, Möller B. Attitude and acceptance of offshore wind farms—the influence of travel time and wind farm attributes. Renew Sustain Energy Rev 2011;15:4223–35.
- [103] Mignon I, Rüdinger A. The impact of systemic factors on the deployment of cooperative projects within renewable electricity production-An international comparison. Renew Sustain Energy Rev 2016;65:478–88.

- [104] Lowitzsch J, Hoicka CE, van Tulder FJ. Renewable energy communities under the 2019 European Clean Energy Package – governance model for the energy clusters of the future? Renew Sustain Energy Rev 2020;122:109489.
- [105] Liebe U, Bartczak A, Meyerhoff J. A turbine is not only a turbine: the role of social context and fairness characteristics for the local acceptance of wind power. Energy Pol 2017;107:300–8.
- [106] Rudolph D, Haggett C, Aitken M. Community benefits from offshore renewables: the relationship between different understandings of impact, community, and benefit. Environ Plan C Politics Space 2017;36:92–117.
- [107] Wolsink M. Wind power and the NIMBY-myth: institutional capacity and the limited significance of public support. Renew Energy 2000;21:49–64.
- [108] Firestone J, Bates A, Knapp LA. See me, Feel me, Touch me, Heal me: wind turbines, culture, landscapes, and sound impressions. Land Use Pol 2015;46: 241–9.
- [109] Bidwell D. The role of values in public beliefs and attitudes towards commercial wind energy. Energy Pol 2013;58:189–99.
- [110] Davis FD. Perceived usefulness, perceived ease of use, and user acceptance of information technology. MIS Q 1989:319–40.
- [111] Vuichard P, Stauch A, Wüstenhagen R. Keep it local and low-key: social acceptance of alpine solar power projects. Renew Sustain Energy Rev 2021;138: 110516.
- [112] Pauly D. Anecdotes and the shifting baseline syndrome of fisheries. Trends Ecol Evol 1995;10:430.
- [113] Kübler-Ross E, Kessler D. On grief and grieving: finding the meaning of grief through the five stages of. Loss Simon and Schuster; 2005.
- [114] Ellis G, Barry J, Robinson C. Many ways to say 'no', different ways to say 'yes': applying Q-Methodology to understand public acceptance of wind farm proposals. J Environ Plann Manag 2007;50:517–51.
- [115] Thayer RL, Freeman CM. Altamont: public perceptions of a wind energy landscape. Landsc Urban Plann 1987;14:379–98.
- [116] Aitken M, Haggett C, Rudolph DP. Wind farms community engagement good practice review. University of Edinburgh; 2014.
- [117] IRENA. 30. Years of policies for wind energy: Lessons from 12 wind energy markets. IRENA GWEC 2013.
- [118] GWEC. Global wind report. 2022.
- [119] WindEurope. Wind industry commitments on community engagement. 2020.
- [120] BBC. Fracking: I'd allow drilling in my back garden, says Jacob Rees-Mogg. 2022.
- [121] Guardian The. Cop27: the future is solar, but it won't be simple. 2022.
- [122] Ziegler L, Gonzalez E, Rubert T, Smolka U, Melero JJ. Lifetime extension of onshore wind turbines: a review covering Germany, Spain, Denmark, and the UK. Renew Sustain Energy Rev 2018;82:1261–71.
- [123] Boyle E, Watson C, Mullally G, Ó Gallachóir B. Regime-based transition intermediaries at the grassroots for community energy initiatives. Energy Res Social Sci 2021;74:101950.
- [124] Klain SC, Satterfield T, Sinner J, Ellis JI, Chan KMA. Bird killer, industrial intruder or clean energy? Perceiving risks to ecosystem services due to an offshore wind farm. Ecol Econ 2018;143:111–29.
- [125] Renn O, Ulmer F, Deckert A. The role of public participation in energy transitions. Cambridge, Massachusetts: Academic Press; 2020.
- [126] Meadowcroft J. What about the politics? Sustainable development, transition management, and long term energy transitions. Pol Sci 2009;42:323–40.
- [127] Bidwell D. Tourists are people too: nonresidents' values, beliefs, and acceptance of a nearshore wind farm. Energy Pol 2023;173:113365.
- [128] Brudermann T, Zaman R, Posch A. Not in my hiking trail? Acceptance of wind farms in the Austrian Alps. Clean Technol Environ Policy 2019;21:1603–16.
- [129] Heffron RJ, McCauley D, Sovacool BK. Resolving society's energy trilemma through the Energy Justice Metric. Energy Pol 2015;87:168–76.
- [130] Bell D, Gray T, Haggett C, Swaffield J. Re-visiting the 'social gap': public opinion and relations of power in the local politics of wind energy. Environ Polit 2013;22: 115–35.